The implications of loss of a partner for older private renters

authored by

Alperhan Babacan, Chris Chamberlain, Grant Cullen, Mike Dockery, Alice Stoakes and Gavin Wood

for the

Australian Housing and Urban Research Institute

RMIT-NATSEM Research Centre

October 2006

AHURI Positioning Paper No. 93

ISSN: 1834-9250

ISBN: 1 921201 21 5



ACKNOWLEDGEMENTS

This material was produced with funding from the Australian Government and the Australian States and Territories. AHURI Ltd gratefully acknowledges the financial and other support it has received from the Australian, State and Territory governments, without which this work would not have been possible.

AHURI comprises a network of fourteen universities clustered into seven Research Centres across Australia. Research Centre contributions, both financial and in-kind, have made the completion of this report possible.

DISCLAIMER

AHURI Ltd is an independent, non-political body which has supported this project as part of its programme of research into housing and urban development, which it hopes will be of value to policy-makers, researchers, industry and communities. The opinions in this publication reflect the views of the authors and do not necessarily reflect those of AHURI Ltd, its Board or its funding organisations. No responsibility is accepted by AHURI Ltd or its Board or its funders for the accuracy or omission of any statement, opinion, advice or information in this publication.

AHURI POSITIONING PAPER SERIES

AHURI Positioning Papers is a refereed series presenting the preliminary findings of original research to a diverse readership of policy makers, researchers and practitioners.

PEER REVIEW STATEMENT

An objective assessment of all reports published in the AHURI Positioning Paper Series by carefully selected experts in the field ensures that material of the highest quality is published. The AHURI Positioning Paper Series employs a double-blind peer review of the full Positioning Paper – where anonymity is strictly observed between authors and referees.

CONTENTS

CC	NTENT	⁻ S	3
LIS	ST OF T	ABLES	5
EX	ECUTI\	/E SUMMARY	6
1	INTRO	DDUCTION	8
1	.1 Back	ground	9
1	.2 Loss	of partner and housing status	13
2		MOGRAPHIC AND HOUSING PROFILE OF OLDER AUSTRALIANS THA	
_		LOST A PARTNER	
2		ographic trends	
	2.1.1	3	10
	2.1.2	The marital history, income and housing tenure profiles of older single persons that have lost a partner	
	2.1.3	Housing tenure and age	
	2.1.4	Marital history and housing tenure	22
2	.2 Hous	ing affordability among older Australians that have lost a partner	26
	2.2.1	Older private renters and Housing Affordability Stress	27
	2.2.2	Home purchasers and Housing Affordability Stress	29
2		mary	
3	NEXT	STAGES IN THE RESEARCH	34
3		of partner and the consequences for housing circumstances: a longituding	
	3.1.1	Key research questions	34
	3.1.2	Approach	34
3		Department of Family and Community Services' Longitudinal Dataset of nefit recipients	36
3	.3 Home	eswest tenancy records and projecting the demand for public housing	38
	3.3.1	Data Description	38
	3.3.2	Preliminary Results - Do older persons experiencing partner loss seek to enter public housing? Are they successful in entering public housing	
	3.3.3	Preliminary findings – How large is the group of older persons that have lost a partner likely to be in the next 10-20 years? What implications mighthis have for public housing demand?	ght
3		are the implications of loss of partner for residential location of older sons?	42
	3.4.1	Data Description	
	3.4.2	Methodology	
	3.4.3	Preliminary Results	
4		ITATIVE RESEARCH AND THE SURVEY INSTRUMENT	
4		tative research: Aims	
-	4.1.1	Conducting the research and interview instruments: How	
	4.1.2	Benefits of the qualitative research: Output	45

	4.1.3	Question themes	45
4.	2 Futur	e housing aspirations	47
5	THE P	POLICY SIGNIFICANCE OF THIS RESEARCH	49
ΑP	PENDIX	<	51
RE	FEREN	CES	62

LIST OF TABLES

Table 1 - The Age Distribution of the Australian Population
Table 2 - Current Marital Status by Age Groups, Persons aged 50 years or Older 1981 17
Table 3 - Current Marital Status by Age Groups, Persons Aged 50 years or Older 2001 17
Table 4 - Married or De Facto in 200118
Table 5 - Marriage Rate ¹ 19
Table 6 - Median Age at Marriage and Remarriage19
Table 7 - Remarriage Rate per 1000 Divorces Granted
Table 8 - Divorce Rates Summary Statistics
Table 9 - Tenure Status of Income Units By Age, 2000-01
Table 10 - Tenure Status of the Population of Older People by Marital Status, 200123
Table 11 - Tenure Status And Marital History Of Older Couples24
Table 12 - Marital History of Older Persons, 200124
Table 13 - Gross Income Distribution ¹ (Defined by Total income unit sample) of older Australian income units by marital status, 2000-01
Table 14 - Median HAR ¹ and Incidence of Housing Affordability Stress (HAS) Among Private Renters Aged 50+: before Commonwealth Rent Assistance28
Table 15 - Median HAR ¹ and Incidence of Housing Affordability Stress (HAS) Among Private Renters Aged 50+: after Commonwealth Rent Assistance
Table 16 - Median HARs and Incidence of Housing Affordability Stress Among Home Purchasers +50
Table 17 - Median HARs and Incidence of Housing Affordability Stress Among Home Purchasers and Private Renters: after Commonwealth Rent Assistance
Table 18 - Median HARs and Incidence of Housing Affordability Stress Among Private Renters: Before Commonwealth Rent Assistance
Table 19 - Median HARs and Incidence of Housing Affordability Stress Among Private Renters: After Commonwealth Rent Assistance
Table 20: Homeswest Tenancy Records – Number of Applicant and Average Waitlist Times for Selected Groups
Table 21: Single Adult (Based on ABS Series B Population Projection)
Table 22: Single Parent Projection of Demand for Public Housing (Based on ABS Series B Population Projection)

EXECUTIVE SUMMARY

This study of the implications of loss of a partner for older private renters is being conducted collaboratively across several AHURI Research Centres, led by the AHURI RMIT–NATSEM Research Centre. The study focuses on how older Australians' (defined as 50+years) housing outcomes are influenced by the loss of a partner through death, divorce or separation. The main focus of the study is the impact on renters rather than owners as the housing circumstances of renters is considered more susceptible to financial stress and uncertainty. However, there are homeowners who lose homeownership status following loss of a partner, and the absence of housing assistance programmes targeted on existing homeowners means that divorcees in this tenure can be vulnerable to housing affordability stress. An important theme of this study is then the important role of losing a partner in determining the housing careers of both older owners and renters. Finally, it should be stressed that most divorces and separations occur well before the partners in a marriage turn 50 years of age. A focus on older Australians ignores the housing career adjustments that divorcees and separated persons make in the initial years following household dissolution.

Section 1 introduces the background and main aims of the study. Section 2 provides a detailed demographic and housing profile of older Australians that have lost a partner. The key findings can be summarised as follows:

- Older single persons who have lost a partner are a rapidly growing segment of the Australian population. There is also an important gender dimension to this trend with the majority of this population segment being female.
- Divorce and separation rates among older Australians are increasing but death
 is becoming a less important cause of household dissolution, as increasing
 longevity means that married couples remain together for an increasing
 proportion of their lives. However the historically high divorce rates and a
 growing number of Australians who choose to never marry has ensured an
 increasing number of single older person households.
- Outright home ownership remains the dominant housing tenure type for older Australians though a significant minority of older Australians are in private or public rental tenures. Older singles that have lost a partner, particularly those divorced or separated, are much more likely to rent.
- Older Australians that have lost a partner are much more likely to have low incomes and be dependent upon income support programmes. They are also more likely to experience housing affordability stress. However, for private renters Commonwealth Rent Assistance plays an important role in bringing levels of housing affordability stress more into line with the levels experienced among other private renters. With no housing assistance to fall back on, divorcees in homeownership are significantly more likely to experience housing affordability stress.

Section 3 outlines several quantitative approaches for the next stages of the research involving analysis of the Household Income and Labour Dynamics in Australia Survey (HILDA), the Department of Family and Community Services Longitudinal Dataset (FaCS) and the Homeswest (Western Australia) tenancy database. The HILDA and FaCS data will be analysed to examine whether the housing circumstances of older Australians that have lost a partner compare unfavourably with other household types. The Homeswest data will be analysed in association with ABS census data to examine older single persons' patterns of entry for into public housing and the implications for future demand and location of public housing for single older persons.

Section 4 discusses the qualitative approaches that will be pursued in the next stages of the research that aim to fill several gaps in knowledge relating to the housing careers of older single persons. This will be achieved with a survey instrument designed to gather information about the impact of losing a partner on specific housing needs, tenure preferences, affordability issues and the role of social networks in influencing the housing decisions of older single persons.

Section 5 concludes with a discussion of the policy significance of this research. We address the policy implications of those demographic trends that result in growing numbers of older single women. In addition the impact of divorce and separation on housing careers for both *younger* and older population cohorts of the Australian population is acknowledged, with recognition that loss of a partner also impacts significantly on younger and middle-age cohorts where the need for housing assistance is also important. Loss of a partner also has a significant impact on homeownership status with those experiencing loss disadvantaged in terms of attaining and retaining homeownership. This also creates policy concerns for the types of housing assistance programmes available following partner loss.

1 INTRODUCTION

Many older Australians are vulnerable to finding themselves in unsatisfactory housing circumstances which can significantly detract from their standard of living. With age, housing requirements change significantly. Yet attachments to the family home, local surrounds and nearby family, financial inflexibility, the need for access to certain services and possibly care arrangements can all mean that people are restricted in their ability to adjust their housing circumstances without incurring significant financial and emotional costs. Compared to home owners, renters are far more likely to face financial stress, as well as uncertainty in their future housing circumstances. This stems from a range of factors, including higher housing costs as a proportion of income; difficulty in maintaining or securing appropriate housing in preferred localities; and a diminishing stock of public rental housing relative to demand coupled with lower levels of assistance for private renters relative to that received by public tenants. Many renters will not have previously achieved homeownership, or will have only been home owners for a limited period in their lives, and therefore will not have accrued the recognised benefits of associated with home-ownership status. The loss of a partner, either through relationship breakdown or death of a partner, is a life effect which can dramatically impact upon housing needs, financial security and the individual's need for external services, family support and care arrangements.

This positioning paper explores the implications of loss of a partner for private renters aged 50 years and over (aged 50+). The scope of the analysis includes singles who turned 50 having previously lost a partner through death, separation or divorce and those who lose a partner after turning 50. The issue is significant not only because of the impact on the individual who has lost their partner, but also because of demographic trends such as the ageing of the population and rising marital separation rates which will find an increasing proportion of the population facing this situation. For the housing industry, it is important that the supply of housing responds to the housing needs of this group, while from a public policy perspective it is important to determine the likely number of such individuals who will require housing related and other assistance, and what is the most appropriate form such assistance should take.

The aim of this project is to understand and measure the effects that divorce, separation, or death of a partner has on the housing and related financial circumstances of older Australians (aged 50+) and in particular those on income support payments by addressing the following questions;

- Do older Australians who have lost a partner compare unfavorably with other household types in terms of housing circumstances and after – housing cost income?
- How are the housing and financial circumstances of older persons affected by loss of a partner?
- Is the probability of losing a partner higher among persons with the following characteristics - Older persons (aged 50+); private rental tenants; income support recipients?
- Do older persons experiencing partner loss seek to enter public housing? Are they successful in entering public housing?
- How large is the group of older persons that have lost a partner likely to be in the next 10 to 20 years?
- What implications might this have for public housing demand? What are the implications of loss of partner for residential location of older persons?
- Do those from diverse cultural and linguistic backgrounds deal with loss of a partner in different ways from the rest of the population?

In this positioning paper we will concentrate on setting out the demographic changes and policy concerns motivating these research questions, and the methods that will be used in our research approach. However, some preliminary findings will presented with respect to the first of the above research questions. The paper is organised as follows: Section one will provide a broad overview of literature relating to the implications of the loss of a partner for older private renters. Section two will present a detailed examination of the demographic and housing profile of older Australians who have lost a partner. Section three will detail future stages of research including the qualitative analysis. This will provide vital information on the impact of the loss of a partner and will also fill existing gaps in research by providing information on the housing needs of older people, issues surrounding tenure, the affordability of housing and the impact of social networks on housing decisions. Section four will provide an analysis of the policy implications of this research.

1.1 Background

The group of primary interest here are defined by three characteristics: being aged 50 and over (50+ years), being a private renter and having lost a partner. It is useful to consider the wider literature on how each of these characteristics, in isolation, impact upon housing circumstances, as well as the more limited body of research assessing the implication of two or all three of these characteristics acting in concert.

The housing requirements of older persons have been analysed extensively within the context of the literature on 'housing careers'. Building on seminal contributions by (Kendig 1984; Kendig 1990), housing careers describe peoples patterns of housing occupation by dimensions such as dwelling type, quality and tenure over the course of their lives. An important determinant of housing careers is the pattern of family formation over the life cycle. A stylised view of the traditional housing career is that individuals initially live within the parental home until they become financially independent and move into rental accommodation. With marriage and child rearing a family home is purchased, and the family will 'ascend' the housing ladder to larger or otherwise superior housing as the family grows and wealth is accumulated. Increasing investment in housing during this stage may also be motivated by saving for retirement, and may take the form of increased investment in the family home or in investment property. In later life trading down or last time sales will occur as the children become independent and leave the family home, reducing the requirements in terms of living space, and as people convert property assets to income streams in retirement. In old age, care and servicing requirements ultimately dictate the form of accommodation individuals can access. The location of housing through these stages will be strongly influenced by career considerations, with access to employment opportunities being critical during prime working ages (Winter and Stone 1999; Clarke et al. 2003; Beer et al. 2006).

Of course, the housing careers of many individuals and families do not follow this stylised path. Analysis of housing careers in the United States by (Clarke et al. 2003) found that nearly a quarter of careers observed over 26 years of data from the Panel Study of Income Dynamics display a move down the housing ladder, such as from home ownership to renting, or from high price rentals to low price rentals, with many in this group likely to be persons who have become divorced or other older single persons reducing their housing consumption. The peak in the housing career is typically attained between the ages of 40 and 50 (Clarke et al. 2003). Income is the primary determinant of what housing is achieved.

A number of demographic trends are thought to be affecting housing careers, including delayed departure from the parental home, marrying and having a first child later in life, longer periods of cohabitation before marriage, higher rates of divorce and separation and more sole parent families. There is some evidence that the impact on housing careers of critical life events relating to family formation, such as marriage and the

arrival of the first child, is diminishing relative to employment and income considerations (see Beer et al. 2006). The idealised housing career is seen as one of positive progression with lengthy periods of stability interrupted by moves into higher quality housing, or at least housing more suited to the individual's or family's requirements. Over much of the housing career horizon home-ownership, as opposed to renting, is seen as the superior state, though it should be noted that owner-occupied housing as a social norm is in fact a relatively recent phenomenon in itself.

Some of the common housing traits of older persons relative to those in their 30s and 40s include lower mobility, diminishing family sizes and reduced emphasis on career and geographical proximity to employment opportunities. Another trend for aged people is towards 'aging in place' in post-retirement housing, rather than in specialist aged care accommodation (Howe 2003; Olsberg and Winters 2005; Beer et al. 2006). Moreover, the housing that is attainable for people in old age is largely determined by their past housing career, most notably whether or not they achieved home ownership and have paid off their mortgage (Kendig and Neutze 1999). Based on 1994 data for Australia, the rate of outright ownership continues to increase with age beyond 50, peaking at just under 80 percent for 65-74 year olds, while the proportion of purchasers decreases. Around 15 percent of persons aged 50 and over are renters, with the balance tipping away from private rental tenancy toward public rental tenancies in old age (Kendig and Neutze 1999). US research indicates that retirement is the critical life cycle point at which preferences for single family detached homes peaks for home owners, after which housing consumption declines (Clarke et al. 2003).

Australians aged 65 years or more have been identified as one of the least mobile groups in the Australian population, and their residential movement is induced more by necessity and less as a result of choice. Generally older Australians move relatively infrequently and when they do, they tend to relocate by moving only a short distance (Jones et al. 2004). Studies completed in Britain, North America and Australia all indicate that older people who did move in later life favoured moving into small scale, purpose built accommodation located in familiar environments and with convenient access to support services, facilities and transport. A link has also been established associating changes in life circumstances resulting from widowhood, separation and divorce with movements by the aged to environments that address social, safety and support needs by providing companionship, supportive services and assistance with the tasks of daily living (Jones et al. 2004).

Although the aged as a whole have a low degree of residential mobility, mobility rates rise with increased age within the group. (Faulkner 2001) cites data from the ABS publication *Older People Australia: A Social Report 1999*, showing 21% of people aged 65-74 and 21% of the 75-84 age group moved residence. This figure rose to 30% for those aged 85-94 and to 37% for those 95 years or older. The very old were thus more likely to move but they were also less likely to have relocated very far, reflecting their very strong affiliation with a particular neighbourhood. It is therefore likely that the increasing numbers of older people moving, particularly those aged over 80 years, will mean that relocation continues to be a prominent housing issue for older Australians (McDonald and Kippen 1999; Faulkner 2001).

An Australian survey of 600 older tenants found them to be agreeable to moving house and changes of tenure, but often maintained a strong attachment to a specific locality. The survey also found that most older private renters expected to move in the next 5 years but were anxious about the prospect of being forced to move due to financial difficulties (Olsberg and Winters 2005). Older people often face numerous obstacles which prevent them from changing residences and moving. This may involve issues of affordability including the financial costs of relocating such as real estate agent and removal expenses. These represent strong financial disincentives especially considering most older Australians are on fixed, low incomes or are pensioners (Faulkner 2001).

A desire to remain in the family home can, from the perspective of the policy community, result in older home-owners occupying more housing than needed given their utilisation of living space, and the wealth 'locked up' in the home. Home owners can afford such a choice in older age because their housing outlays are typically very low. They are also encouraged to hold wealth in owner occupied housing by favourable tax treatment and exemption of the family home under the asset test for income support programmes (ISP), including the age pension. Renters are far less likely to have the choice of 'aging in place', and more likely to have to respond to the vagaries of economic pressures as they have both higher housing costs and lower incomes (Kendig and Neutze 1999) Renters also have greater mobility, defined in terms of the propensity to change residences, than homeowners, and mobility is the main mechanism through which households bring their consumption into line with their needs (Clarke et al. 2003).

For these and various other reasons, a significant proportion of people find themselves in less than satisfactory housing in old age, and one of the major causes of this is interruption to the housing career through family dissolution. Australian research based on 74 interviews with older clients and managers of the Assistance with Care and Housing for the Aged (ACHA) program revealed that despite an overall increase in the wealth of older Australians over the past twenty years, a significant proportion of the group aged 65 years or older are in a vulnerable housing situation and may find themselves facing the prospect of homelessness (Morris et al. 2005). This was found to be particularly pertinent for older, low income private rental tenants who depend on the government for their income. The research also indicated that the loss of a partner through death, divorce or separation is a common contributing factor, because it leaves an older individual less able to find suitable accommodation and in some cases, can cause homelessness. Older, single Australians are thought to be especially vulnerable as many have minimal or no family and social networks and often can not rely on the housing market or government sector bodies to assist them in securing affordable housing. Their predicament is further exacerbated by the fact that older people living alone are more likely to require social and community care services, such as home help and community care services.

At a 2004 AHURI conference, it was pointed out that older renters, on fixed low incomes, were less able to effectively compete for the declining supply of inexpensive, private rental housing. Further, rent assistance payments were insufficient to enable older renters to compete in the rental market, and many needed other support to find secure and stable housing that also met their needs in terms of location and utilities (AHURI 2004). Drawing on figures from AHURI (1996), Howe (1992) and Kendig and Neutze (1999) argue that despite having low incomes, older Australians who privately rent accommodation generally have significantly high renting costs and have been identified as among those in greatest housing need. According to 1995 AHURI figures, private tenants made up two thirds of the older households whose renting costs exceeded 30% of their gross income and 25% of older private renters paid more than half their income on rent. The percentage of older couples or singles paying 25% or more of their income on rents was 68% for private tenants and 14% for public housing tenets. ABS data reveals that older couples renting privately were paying on average 30% of their incomes on housing expenses while single, older Australians were paying 50%, confirming that they are among those in greatest housing need

The desire of some single, older people to enter public housing is understandable considering that it provides them with both security of occupancy and comparatively low renting costs. Indeed Australians aged 65 years or more are one of the key target groups for the public rental sector (Jones et al. 2004). Public housing targeted on the elderly, such as higher density housing developments zoned specifically for aged residents, has been found to offer other advantages including more informal support, increased social contact and improved delivery of community services (Brooke et al.

1998). Research completed by AHURI also indicates that older Australian renters in public housing have lower housing costs than those who privately rent (Kendig and Neutze 1999). Presently only a small minority of older Australians live in public housing, but given the small size of the tenure they make up a significant proportion of the public tenant population (Jones et al. 2004). According to ABS data approximately 5.3% of the population aged 65 years or older live in public rental housing (Faulkner 2001). Increased demand for public housing has resulted from a rise in the numbers of single households who can't afford to buy their own home or remain renting privately (Smyth and Weston 2000).

Significant reductions in Commonwealth funding over the past few years has resulted in consequent decreases in the supply of public housing. Attempts by State Governments to compensate for the funding shortfall have had minimal impact in increasing the stock of public housing (House of Representatives Standing Committee on Health and Ageing 2005). The diminishing supply has therefore meant that the level and range of public housing in Australia is insufficient to satisfy demand and there is increasing competition for the available places that do exist (AHURI 2004). This is reflected in figures indicating that access to publicly provided housing has declined. It is also likely to translate into fewer long term renters being able to move into public housing when they retire. Evidence of this trend is apparent from a fall in the share of older age (65+) renters, renting in the public sector (Smyth and Weston 2000; AHURI 2004; House of Representatives Standing Committee on Health and Ageing 2005).

Some authors have also reported that the changing nature of the housing market has meant that single, older Australians are finding it increasingly difficult to find affordable, adequate and secure accommodation (Morris et al. 2005). The housing industry has been slow to adjust to changing social trends and the imminent rise in the number and proportion of older Australians. New housing construction continues to be dominated by family housing needs and fails to offer the range of choices that are attractive and affordable for many older individuals. Furthermore, modified State urban planning schemes have tended to promote high density housing, mostly in expensive, inner city locations and in multi-story dwellings, not suited to the needs of older people (AHURI 2004). In more recent times attempts to increase the supply of public housing have concentrated on establishing public-private sector partnerships with the aim of attracting private sector investment to provide more affordable housing options for older persons (Jones et al. 2004).

Another factor which has been identified as contributing to the affordable housing dilemma faced by many older Australians is the fact that government housing support for those on low incomes has changed substantially in recent years. Rent assistance has become the primary form of housing assistance, replacing public housing that had been the dominant type of housing assistance until the 1980s. (House of Representatives Standing Committee on Health and Ageing 2005). Many commentators claim that the maximum payment for rent assistance is relatively low when compared to rent rebates for public tenants (Feijten and Mulder 2005), and is insensitive to required variation in rents (Wood et al. 2005). Rent Assistance is thus considered to be of limited help and largely ineffective in overcoming the affordability difficulties faced by older renters. Further, older people face other difficulties with regard to accessing rental housing, including the need to have housing close to stores, public transport and other services (Fiedler 1999; AHURI 2004).

The housing challenges facing older, single persons are exacerbated from a social and policy perspective because the size of this group will increase significantly in the next couple of decades. This research project provides projections, out to 2025 - of the number of Australians aged 50 and over who have lost a partner, as well as estimating the implications of this for public housing demand. Previous research has documented changing demographic trends contributing to the growth in this group. According to AHURI figures, it is estimated that by 2021, 18% of Australia's population will be aged

65 years or above, up from 12% in 2001. Based on these figures, a person of this age group is likely be a resident in approximately 4 in 10 households (AHURI 2004). It is estimated that 28% of older Australians currently live alone and the ABS predicts this figure to substantially increase by 2021 (Jones et al. 2004). The ABS and AHURI statistics show that roughly 90 % of aged persons live in private housing and that around 10-15% of this group are renters (AHURI 1996; Faulkner 2001). Slightly more than half of these older renters rent in the private market while the remainder are predominantly in public housing. The proportion of older age renters is likely to increase in the future as more become long term renters due to the fact that maintaining home ownership or making a transition into home ownership is becoming increasingly difficult for lower income Australians (Jones et al. 2004). This is supported by research which estimates that there will be a rise in the absolute number of people entering old age as renters and also more older Australians looking to rent in the private market (AHURI 2004).

Studies also indicate that the rise in single person households will continue and accelerate over the next two decades, across all age groups. These trends are therefore likely to generate greater demand for additional and more diverse types of housing in Australia, especially within major capital cities. One example of the implications is the increase in demand predicted for housing that incorporates various forms of support to cater for the needs of less mobile and more frail older Australians (AHURI 2004). This is particularly relevant considering the imminent rise in the numbers of people with severe core activity restrictions and older women living alone that has been predicted to occur. All projections thus point to rapid growth in demand for public housing by older Australians, with the types of property in demand not well matched to the existing public housing stock. Funding cutbacks however are likely to limit the capacity of the public housing system to meet the rising demand. Consequently the accessibility of affordable and stable housing for older renters is likely to remain a prominent issue, which will arguably gain even more importance in the future (McDonald 2003; AHURI 2004; Jones et al. 2004).

1.2 Loss of partner and housing status

A recurring theme from studies that we have reviewed is the important contributory role of divorce or separation in disrupting housing careers (Dieleman et al. 1995; Clarke et al. 2003; Morris et al. 2005). The immediate impact of separation is an increase in demand for housing because one household 'dissolves' and is often replaced by two households. Not all relationship dissolutions result in additional households, as upon leaving one relationship a significant number of people move in with a new partner or move back into their parents' homes. UK research has shown that each dissolved marriage leads to 1.53 household units seeking accommodation (McCarthy and Simpson 1991). Following separation it is difficult for a single income household to move into home ownership. Evidence from the US (Dieleman et al. 1995) suggests that both single person households and sole parent families have a very low likelihood of purchasing a home. If they are already in home ownership, these families also have a greater propensity to move from owning to renting, due to lower income and the absence of a stable relationship which fosters commitment to purchasing and repayment of a mortgage.

In old age the loss of a partner can have more pronounced effects (Dieleman et al. 1995). Death of a spouse is one of the most common causes of changes in living arrangements among individuals in later life. In one UK study it was found to be responsible for 80 % of all the transitions from a couple household to a single person household (Fiedler 1999; Evandrou et al. 2001). Several studies have shown that on average, people residing with others either as a result of marriage or cohabitation, live in higher quality housing, than those who live in one-person households (Dieleman et al. 1995). Interruptions of partnerships due to widowhood, divorce or separation have

been found to have both short term and lasting negative consequences, causing an overall decline in the quality of housing for individuals (Boheim and Taylor 2000; Fischer and Malmberg 2001; Feijten and Mulder 2005).

Results from an Australian Institute of Family Studies telephone survey of 137 divorced, older Australians showed that the loss of a partner due to separation or divorce often caused economic crises. One leading reason for this outcome is that the ex - couples income, which supported a joint household, was generally insufficient to meet the expenses of newly formed single households (Smyth and Weston 2000; Fischer and Malmberg 2001). This is consistent with the findings of a recent National Longitudinal Survey conducted in the US, which indicated that individuals experience a significant decrease in wealth post separation from their partners (Zagorsky 2005). There is debate in the literature as to whether it is males or females who, on average, do better out of divorce settlements. What is clear is that divorce and separation causes significant hardship for both genders (Beer et al. 2006). As women more often retain custody of dependent children, there is a tendency for the mother to remain in the family home to try to minimise disruption, but this can be soon followed by a transition down the housing ladder. Overall, however, older women have been found to be consistently more disadvantaged than men, both financially and in terms of housing, following separation from their partners (Smyth and Weston 2000; Sevak et al. 2003/4; Zagorsky 2005). The results of the AMP-NATSEM Income and Wealth Report, which examined the income and wealth of Australian retirees, appears to support this assertion. In the report it was found that the poorest 25% of single females had wealth that was approximately one-third that of the poorest 25% of single males (AMP - NATSEM 2002).

Several studies have reported that divorce often causes individuals to move from owner occupied to rental housing and that divorced men and women are more likely to move into public housing than other singles (McCarthy and Simpson 1991; Dieleman et al. 1995; Feijten and Mulder 2005). This is supported by the results of a study based on data from the Household, Income and Labour Dynamics in Australia (HILDA) survey, and the Australian Bureau of Statistics' (ABS) Survey of Income and Housing Costs (SIHC) (Flatau et al. 2004). This study found that amongst all age cohorts in the mature age spectrum, those who are divorced or separated from their partners have significantly lower rates of homeownership (when compared to continuously married couples), regardless of age. Further evidence exists in the finding that the majority of individuals who divorced in mid-life have lost and not regained homeownership (Kendig and Neutze 1999).

In the next section we provide demographic and housing profiles of older Australians that offer a more up-to-date profile of their circumstances. It goes beyond previous studies by describing the housing affordability position of this group taking Commonwealth Rent Assistance into account. In addition it exploits the retrospective household histories contained in the Household Income and Labour Dynamics Survey. This allows a richer analysis of housing circumstances by marital history.

2 A DEMOGRAPHIC AND HOUSING PROFILE OF OLDER AUSTRALIANS THAT HAVE LOST A PARTNER

Numerous demographic trends that are apparent in the Australian context and have potentially profound implications for the future demand for housing assistance. One of these trends is marked growth in the number of single person households, and this has been particularly so among older persons. It partly reflects the growing importance of divorce and separation, and an ageing of the population that is accompanied by a corresponding increase in the number of single persons that have lost a partner due to bereavement. This trend has a wider social policy context since older people living alone are more likely to be in receipt of social and community care services, such as home help and community care services (Evandrou et al. 2001). In view of this it seems likely that such households will also form an increasingly important source of demand for housing assistance, including public housing. In this section we sketch in the demographic background by examining trends over the last twenty years using Australian Bureau of Statistics (ABS) population estimates. This is followed by a description of the housing tenure profile of older Australians with different marital histories, and an analysis of their relative position in the nation's income distribution. We ask whether older Australians that have suffered household dissolution experience disadvantaged housing and financial circumstances. Finally we present a descriptive analysis of housing affordability stress among older persons, with a particular focus on benchmarking the position of single person households that have lost a partner due to death of a spouse, divorce or separation. This last exercise addresses the first key research question posed in our proposal: are older Australians who lose a partner suffering a financial loss that increases the need for housing assistance?

2.1 Demographic trends

Table 1 examines changes in the Australian population distribution over the period 1981 – 2001 using Australian Bureau of Statistics (ABS) population estimates (ABS 2006) (ABS cat. no.3105.0.65001). In 1981 people aged 50 years or older constituted some 32 % of the total population aged 15 years or older. By 2001 this share had increased to 36 %. There are three main trends evident on inspection of table 1

- Between 1981 and 2001 the Australian population grew by over one-third; from 11.4m to 15.4m
- Population shares in the younger age bands 34 years and below systematically declined;
- Population shares for the age bands 40-54 years and 70 years and above increased.

Table 1 - The Age Distribution of the Australian Population

Age group	1981 – Number of Persons	Share of Total Population in 1981 (%)	2001 – Number of Persons	Share of Total Population in 2001 (%)	Percentage Point Change in Share of Total Population (%)	Percentage Change in Number of Persons (%)
15-19	1,288,994	11.3	1,352,745	8.8	-2.5	4.9
20 - 24	1,333,401	11.7	1,302,412	8.4	-3.2	-2.3
25 - 29	1,253,635	11.0	1,407,081	9.1	-1.8	12.2
30 - 34	1,228,574	10.7	1,466,615	9.5	-1.2	19.4
35 - 39	1,073,243	9.4	1,492,204	9.7	0.3	39.0
40 - 44	865,803	7.6	1,479,257	9.6	2.0	70.9
45 - 49	748,322	6.5	1,358,594	8.8	2.3	81.6
50 - 54	766,107	6.7	1,300,777	8.4	1.7	69.8
55 - 59	746,262	6.5	1,008,799	6.5	0.0	35.2
60 - 64	636,152	5.6	822,024	5.3	-0.2	29.2
65 - 69	542,837	4.7	682,513	4.4	-0.3	25.7
70 - 74	418,048	3.7	638,380	4.1	0.5	52.7
75+	537,883	4.7	1,114,641	7.2	2.5	107.2
Older Persons 50+	3,647,289	31.9	5,567,314	35.9	4.2	0.0
Total	11,439,261	100.0	15,426,042	100.0	0.0	34.9

Source: ABS cat. no. 3105.0.65001

The Australian population is ageing with the number of persons aged 75 plus years more than doubling over the twenty year period. As table 1 demonstrates the picture is complicated by the baby boomer generation. In 2001 the age cohort born between 1946 and 1960 belonged to the age bands 40-54 years, and over the past 20 years this baby boomer segment of the Australian population has increased its share of the total population. This group are now approaching early retirement age but by 2025 the surviving baby boomers will be 65-79 years of age and will be responsible for acceleration in the ageing of the Australian population. In the past 20 years the ageing process has been restrained by the age cohort born between 1931 and 1945 who belonged to the age bands 55 – 69 years in 2001. Over the past 20 years this segment of the Australian population has been contracting. By 2025 surviving members of this cohort will be 80+ and their influence on the age distribution will be waning by comparison to that of the baby boomer generation.

There is a belief that family relationships have become less permanent as separation and divorce has become more common (Winter and Stone 1999). There is also a view that people are now less likely to enter into marriage and are more prepared to live independent lives in single person households (Winter and Stone 1999). We explore whether these trends are evident among the 50+ older age group, where in table 2 the Survey of Income and Housing Costs (SIHC) is used to classify persons into 5 groups according to their marital histories in 1981, and in table 3 the ABS census of population is drawn on to measure the same categories in 2001.1

For the 50+ age group as a whole some 70% of persons were married in 1981, but by 2001 this share had declined to 65%. The corresponding share of persons who had experienced the loss of a partner (through separation, divorce or death) increased from

¹ It is not possible to use the ABS census for 1981 because the ABS's Australian Demographic Statistics series does not contain marital status by age group.

24% in 1981 to 29 % in 2001². However, there has been significant compositional change within this group. *The share of widowed persons has declined from 19% to 16%, whereas the share of separated or divorced persons has increased from 6% to 13%.* Though there is a popular perception that people are becoming less likely to marry the share of the never married remains very small at 5.6% of the 50+ age group in 2001.

These overall trends mask some important differences across generations of older Australians. Among those in their 50s in 1981, and therefore born between 1922 and 1931, marriage shares are high at around 80%.³ But among the corresponding age group in 2001, who were born between 1942 and 1951, the marriage share has declined to 72%. This decline is largely due to the higher propensity to divorce and separate of this (largely) baby boomer generation who were beginning to enter the older 50+ age bracket by 2001.

Table 2 - Current Marital Status by Age Groups, Persons aged 50 years or Older 1981

Age groups	Married %	Separated %	Divorced %	Widowed %	Never Married %	Total %	Population (Number)	Share of Total Person aged 50+ years %
50 - 59	81.3	2.0	4.8	6.4	5.6	100	1,502,027	42.4
60 - 69	72.1	1.7	3.4	17.3	5.5	100	1,166,334	32.9
70+	47.7	0.9	2.0	42	7.4	100	876,200	24.7
Total 50+	69.9	1.7	3.6	18.8	6.0	100	3,544,561	100

Source: Confidentialised Unit Record Files of the ABS (1981/82), Income Distribution Survey (more recently renamed Income and Housing Costs Survey)

Table 3 - Current Marital Status by Age Groups, Persons Aged 50 years or Older 2001

Age Groups	Married %	Separated %	Divorced %	Widowed %	Never Married %	Total %	Population (Number)	Share of Total Persons age 50+ years %
50 - 59	72.1	4.7	13.5	3.3	6.4	100	2,219,220	41.3
60 - 69	71.2	3.2	9.9	10.6	5.0	100	1,449,594	26.9
70+	50.4	1.6	5.0	37.8	5.2	100	1,710,291	31.8
Total 50+	65	3.4	9.8	16.2	5.6	100	5,379,105	100

Source: Confidentialised Unit Record Files of the ABS (1981/82), Income Distribution Survey (more recently renamed Income and Housing Costs Survey)

On the other hand the marriage share among the 70+ age bracket has increased due to lower mortality rates that are enabling larger numbers of older Australians to live longer with their partners. Those Australians in the 70+ age bracket in 1981, and therefore born before 1912, had a marriage share of 48%. But this marriage share has increased to 50% for those aged 70+ in 2001 (and therefore born before 1932)⁴. There

17

² These figures are the row sum of the share of persons who have lost a partner because of separation, divorce or bereavement in tables 2 and 3.

³ We use the term shares when referring to stock measures (e.g. the proportion of the population that is married) and the term rates when referring to flow measures (e.g. the number of persons that married in the last year per 1,000 residents).

⁴ Note here that marriage share is the proportion of the population or population subgroup that are *currently* living in a married couple household.

is a corresponding fall in the share of widows. In short it would seem that the propensity to marry has not fallen among the cohort of Australians that reached the age of 50+ years by 2001, but they are much more likely to have lost a partner as a result of divorce and separation as compared to their counterparts in 1981. They are also much less likely to be widowed as a consequence of increasing longevity of the older segment of the population. There are offsetting trends; those married couples that avoid separation and divorce will live longer as couples because of falling mortality rates. On the other hand fewer married couples survive as the propensity to divorce and separate has increased. Only 4% of 50+ Australians were divorced in 1981; by 2001 this had reached 10% of the 50+ age bracket. The share of separated persons increases from 2% to 3% over the same time frame. Population estimates indicate that older 50+ persons separated or divorced persons increased from 184,000 in 1981 to 705,000 in 2001, a nearly four fold increase in this subgroup of older Australians. We are therefore witnessing a rapidly increasing number and share of older persons who are living alone without a partner because of divorce and separation. Though widows are still the largest group overall, it is divorce and separation that is driving change in the level and composition of 50+ older persons that have lost a partner.

This conclusion is subject to a qualification. The formation of de facto partnerships is not taken into account in the above analysis. However, as documented in table 4 this living arrangement is much more common among younger Australians, with the incidence of de facto partnerships peaking at 17.6% of 25-29 year olds. Among 50+ years old Australians it declines from 5.2% of persons aged 50-54 years to only 0.6% of persons 75+ years. It therefore seems reasonable to conclude that Australia is experiencing a rapid increase in the number of older Australians that are living alone. In view of the importance of divorce and separation to this conclusion, we now take a closer look at trends in divorce and marriage rates.

Table 4 - Married or De Facto in 2001

	Married in a registered marriage	Married in a de facto marriage	Not married	Total
	% Share of persons in age band	% Share of persons in age band	% Share of persons in age band	% Share of persons in age band
15-19 years	0.4	1.9	97.7	100
20-24 years	8.4	13.1	78.5	100
25-29 years	34.7	17.6	47.7	100
30-34 years	55.3	13.1	31.6	100
35-39 years	64.4	9.6	26.0	100
40-44 years	67.7	7.6	24.7	100
45-49 years	69.5	6.4	24.0	100
50-54 years	71.5	5.2	23.3	100
55-59 years	72.6	3.9	23.5	100
60-64 years	72.3	2.5	25.2	100
65-69 years	69.6	1.5	28.9	100
70-74 years	63.4	1.0	35.6	100
75+	45.7	0.6	53.7	100

Source: Census 2001 (usual resident profile)

2.1.1 What are the trends in rates of marriage, remarriage and divorce?

While the number and share of divorced people is increasing in the 50+ year age group it is important to note that marriages and divorces tend to occur at younger ages. Thus changes in divorce rates and marriage rates will have lagged effects on the marital composition of the 50+ age group. Tables 5 to 8 examine population, marriage,

remarriage and divorce rate summary statistics. The rate measures are the number of persons experiencing an event (marriage, for example) in a calendar year divided by the total number of adults (aged 15 years or older).

Table 5 - Marriage Rate¹

	1983	1993	2003
Marriage rate ² : All adults	7.5	6.4	5.4
Males			
Marriage rate : among adult males 50+ years	3.7	3.5	3.5
Females			
Marriage rate: among adult females 50+ years	1.9	1.9	2.0

Note 1: rates are per 1000 adults. The male rates are per '000 males, and the female rates are per '000 females.

Note: 2 the male (female) remarriage rate per '000 marriages has increased from 238 (225) in 1983 to 249 (238) in 2003.

Source: Australian Demographic Statistics Sept Q 2004 (ABS 3101.0).

In the period 1983-2003 the marriage rate (see table 5) declined from 7.5 (per thousand adults) to 5.4 (per thousand adults). There has therefore been a substantial 28 % decline in the propensity of Australian adults to enter marriage. This decline is apparent in almost all age groups, an exception being the rate for the older 50+ female age group. However, in the 50+ age group the marriage rate is low, a finding that is to be expected given a median age at marriage that is now just over 30 years.

Table 6 examines trends in the median age at marriage and remarriage. By 2003 males and females had typically delayed marriage by 5 years as compared to 1983, and the median age of marriage had reached around 31 years for males and 29 years for females. Among males and females whose first marriage dissolved due to divorce, the median age at remarriage has also increased, and for this group the delay is even more pronounced. Widows who remarry do so in late middle age and this tendency has become more pronounced.

Table 6 - Median Age at Marriage and Remarriage

	1983	1993	2003
Males			
Median age (all) marriages	26.4	28.8	31.2
Median Age at remarriage: Divorced	36.5	40.4	43.6
Median Age at remarriage: Widowed	59.7	62.0	62.6
Females			
Median age (all) marriages	23.9	26.4	29.1
Median age at remarriage: Divorced	33.4	36.8	40.2
Median Age at remarriage: Widowed	52.3	52.8	55.8

Source: see table 5.

Changes in remarriage rates are critical from a policy perspective, because the adverse consequences of divorce or partner bereavement could be reversed on remarriage. A reduced propensity for divorcees to remarry is apparent in table 7. The remarriage rate per 1000 divorces granted has fallen from 852 in 1983 to 672 in 2003. Thus the propensity of <u>divorcees</u> to remarry is in decline.⁵ This is an important

⁵ There is qualification here. Remarriages include those involving widows and so these trends could reflect a change in the remarriage rate among widows. However, remarriages involving widows are small in

observation as it implies a *net* growth in the number of single person households that have lost a partner due to divorce, and the growth will be increasingly apparent among the 50+ age group as the influence of older generations (whose marriages were less prone to divorce) wanes

Table 7 - Remarriage Rate per 1000 Divorces Granted

	1983	1993	2003
First marriage one partner, other partner remarrying	484	436	370
Remarriage both partners	368	334	302
Total	852	770	672

Source: See table 5.

The falling rate of marriage has helped stabilise divorce rates (per 1000 adults). The divorce rate in 1983 was 2.8 and fell marginally to 2.7 by 2003, but the stability masks different trends across age cohorts (see table 8). Among younger male and female Australians in the age brackets 30-34 and 35-39, divorce rates have declined, though the decline has been from relatively high rates of between 14 and 15 per 1,000 males (or females). In older age brackets the divorce rate has increased. This trend is in part a product of delayed marriage so that the median age of divorce has increased in parallel with the increasing median age of marriage.⁶

Table 8 - Divorce Rates Summary Statistics

	1983	1993	2003			
Divorce rate (per 1000)	2.8	2.7	2.7			
Divorce rate age specific	Husband					
30-34	15.3	13.1	10.8			
35-39	14.2	13.1	12.6			
40-44	12.5	12.8	12.6			
45-49	9.9	10.8	11.7			
50-54	7.2	8.4	9.5			
50+	2.7	2.7	3.6			
Divorce rate age specific	Wife	Wife				
30-34	15.0	13.9	13.1			
35-39	13.3	13.2	13.1			
40-44	10.9	11.5	12.3			
45-49	8.2	9.4	10.5			
50-54	5.2	5.9	7.3			
50+	1.4	1.3	1.9			

Note: The age specific rates are per 1000 males or females respectively aged 15 years or older. *Source*: see table 5.

As the Australian population ages we can expect household composition in the 50+ category to be increasingly affected by the choices that Australians under the age of 50 are making concerning marriage and divorce. It is clearly the case that this latter group of Australians are not as prepared to enter into marriage as their counterparts were 20 or more years ago. However, those that do enter marriage are more likely to divorce,

number and unlikely to alter conclusions based in table 7. According to ABS 3101.0 the male (female) remarriage rate (per '000 marriages) among widows was only 20 (22) in 2003.

⁶ The median age at divorce has increased from 36 years in 1983 to 43 years in 2003 for males and from 34 years to 40 years over the same time frame for females (ABS 3101.0).

and those that get married are also more prepared to 'churn' in and out of marriage.⁷ The increasing share of 50+ persons that are divorced also reflects the fact that not all divorcees remarry. As older Australians live longer than their counterparts twenty or more years ago, the share of widows in the +50 year age group have fallen, and can be expected to fall further given declining marriage rates among younger Australians. The implications of these demographic trends for tenure profiles and the relative income positions of older Australians with different marital histories are now explored.

2.1.2 The marital history, income and housing tenure profiles of older single persons that have lost a partner

This sub - section uses the Household, Income and Labour Dynamics Survey (HILDA) to profile the socio-economic characteristics of older Australians and relate these to their marital histories. The latter are defined to include all income units⁸ that contain at least one person 50+ years of age. HILDA is a valuable data source because it contains retrospective information about family histories that allow us to precisely identify 'pathways' preceding and following loss of a partner. A second important reason for using HILDA is that AHURI-3M, the microsimulation model of the Australian Housing Market, is being redesigned so that it can be applied with HILDA as a data source.⁹ The model contains a tax-benefit simulator that allows measurement of housing assistance, and housing affordability measures before and after taking housing assistance into account.

We begin by comparing the housing tenure profile of older (50+years of age) Australians with that of their younger counterparts. We then explore the marital histories of older Australians, and how these relate to housing tenure outcomes. We find that older singles that have lost a partner are much more likely to be renters, and are three times more likely to rent in public housing than couples. This prompts a more detailed investigation of the demographics and incomes of older singles who have lost a partner. Section 2 is concluded by a presentation of our findings on the housing affordability position of older Australians, and in particular older singles that have lost a partner.

2.1.3 Housing tenure and age

Table 9 shows that in 2000-01 two-thirds of older (50+) income units are outright owners. This is a stark contrast with younger income units where outright owners constitute only a 16 percent share of all income units. Not surprisingly, younger income units have a higher propensity to rent, with 36 percent of this group opting for private rental, compared to only 10 percent of those aged 50 and over. Whilst home purchase is the second most common tenure for both age groups, a higher proportion of younger income units are purchasers (33 percent compared to 14 percent). This is to be anticipated, since older Australians are typically outright owners when they reach their mid - 50s (see ABS 2004). Evidence suggests this tenure profile may change in the future, with indications that younger households are delaying entry into home ownership (Yates 2000). The potential outcome of this trend is an increase in the number of older Australians encumbered with a mortgage as their retirement

⁷ The male (female remarriage) rate per <u>'000 marriages</u> has increased from 238 (225) in 1983 to 249 (238) in 2003.

⁸ Income units are defined as adults and their dependants among whom economic resources are nominally shared. A married couple with a 6 year old daughter are an income unit, but a married couple who have their 28 year old employed son living with them form two income units but one household. Rates of homeownership are always lower when calculated using income units rather than households as the unit of measurement.

⁹ See Wood, Watson and Flatau (2006) where the model is applied to analyse first home owner grants.

¹⁰ Older income units are defined as those where at least one adult is 50+ years of age. Younger income units are then those where all adults are younger than 50 years of age.

approaches. The implications will be particularly severe for older single households, who may face greater financial difficulties meeting mortgage repayments given economies of scale in housing consumption.¹¹

Table 9 lists both sample numbers, i.e. those surveyed in HILDA, and population estimates. There are two important points to bear in mind. Firstly, as cautioned in our response to the project brief, the sample of older renters is 499 income units, a much smaller number as compared to the 2494 older homeowner income units that are sampled by HILDA. Particular sub-groups of older renters (e.g. widows) will then have small sample numbers, and caution is warranted when interpreting findings. Secondly, tenure shares based on sample numbers and population estimates are very similar for the 50+ years age group. Consider the share of outright owners for instance. Both the sample and population estimates of their tenure share is 65.8%. Subsequent tables report sample numbers only.

Table 9 - Tenure Status of Income Units By Age, 2000-01

Tenure	Sample				Population			
	50+ Ye	50+ Years		15-49 Years		50+ Years		
	N	%	N	%	Number ('000)	%	Number ('000)	%
Outright owner	2,067	65.8	885	16.9	2,089	65.8	842	16.2
Owner purchaser	427	13.6	1,794	34.2	432	13.6	1,716	33
Private Renter	316	10.1	1,823	34.8	315	9.9	1,879	36.1
Public Renter	183	5.8	2,257	4.9	194	6.1	2,501	4.8
Rent Free	147	4.7	485	9.2	147	4.6	514	9.9
Total	3,140	100.0	5,244	100.0	3,176	100	5,202	100

Note 1: The sample includes responding income units in two age groups: income units that are aged under 50 (that is where each adult is aged 15 to 49) or income units where at least one person is 50 and over (50+ years). The two groups includes couples (married or de facto), single (and have lost a partner), and never legally married. The analysis uses the HILDA responding person cross-section population weight to translate sample numbers into population estimates. Cases with \$0 or negative income have been excluded. The sample also excludes people who are legally married but not living with their partner or not stated (20 cases) and income units with an uncategorised tenure status (20 cases).

Source: HILDA Wave 1

2.1.4 Marital history and housing tenure

Home ownership is often viewed as an investment, providing financial security and offering greater lifestyle choices (Olsberg and Winters 2005). Home ownership is a dominant feature of the tenure status of older Australians, as is clear from table 10, which examines the population of older Australians by tenure and marital status. Older couple households have the highest rates of home ownership, with 71 percent of income units outright owners, and 17% still paying off a mortgage. A smaller 60% of singles who have lost a partner are outright owners, and only 10 % are still paying off a mortgage. A higher proportion of singles who have lost a partner rent privately (14%), compared to couples (7 %) and those people who have never married (12 %). Table 10 also indicates that older singles who have lost a partner are more likely to enter public housing; 10% of this group rent from a state housing authority as compared to 7% of never married older singles and 3% of older couples.

¹¹ Both couples and singles need certain amenities, such as kitchen and bathroom facilities, and are the source of economies of scale that make it cheaper (on a per capita basis) for couples to purchase or rent housing. A recently released Citibank (2006) survey finds that 20 % of Australians under 45 years of age expect to be meeting mortgage repayments in their retirement years.

Among older Australians who have lost a partner there are important differences between widows on the one hand, and the divorced and separated on the other hand. Only 12% of widows are renters, but 3.5% of the divorced and separated are renters. There is a correspondingly stark difference in rates of homeownership, with 76% of widows in homeownership but only 62% of the divorced and separated in homeownership and almost one third of the latter still have a mortgage. We comment on the policy significance in section 5 below.

Table 10 - Tenure Status of the Population of Older People by Marital Status, 2001.

Tenure	Coupled ¹		Single, lost a partner ²		Never married ³		Total	
	N	%	N	%	N	%	N	%
Outright owner	1267	70.6	693	60.3	107	54.6	2067	65.8
Owner purchaser	299	16.7	112	9.7	16	8.2	427	13.6
Private Renter	131	7.3	162	14.1	23	11.7	316	10.1
Public Renter	50	2.8	119	10.3	14	7.1	183	5.8
Rent Free	47	206	64	5.6	36	18.4	147	4.7
Total	1794	100	1150	100	196	100	3140	100

Note 1: Couple income units are defined as those who are legally married or couples in a de facto relationship at the time of interview.

Note 2: Single income units are defined as individuals who are separated, divorced or widowed at the time of interview.

Note 3: Never married income units are those who have never been legally married at the time of interview, but may have been in a de facto relationship in the past.

Source: Authors calculations from Wave 1 of HILDA

Table 10 clearly indicates that loss of a partner adversely affects your chances of being a homeowner. Many will subsequently form new relationships but will their homeownership prospects 'bounce back' such that they become indistinguishable from the continuously married couples? Table 11 compares the tenure arrangements of older continuously married couples with those older couples where at least one partner has been in a previous marriage or de facto relationship. It is clear from table 11 that outright home ownership is highest amongst continuously married couples (76%). By contrast 62% of repartnered married couples and less than half of repartnered de facto couples (45%) are outright owners. It appears that marriage dissolution has adverse impacts on the propensity to achieve outright ownership regardless of repartnering. Also noticeable is the higher share of owner purchasers among repartnered married (23%) and de facto (34%) couples as compared to the continuously married (13%).

The findings reported in table 11 are important because they suggest that couple households formed as a consequence of repartnering are more likely to be approaching retirement (or in retirement) with outstanding debt or in rental housing tenures. Repartnering does not necessarily put divorcees and widows back on an equal footing with the continuously married. It seems likely that household dissolution prompts housing equity withdrawal that is not offset by equity injection on re-partnering. Though the brief for this study was written with an emphasis on older singles that have lost a partner, the housing affordability and demand for housing assistance position of those that repartner might also be cause for concern.

Table 11 - Tenure Status And Marital History Of Older Couples

Tenure	couple-	married both in first	partner previously married		ne couples- one		Total	
	Number	%	Number	%	Number	%	Number	%
Outright owner	969	75.7	234	61.7	50	45.5	1253	70.8
Owner purchaser	167	13.0	89	23.5	38	34.5	294	16.6
Private Renter	74	5.8	38	10.0	16	14.5	128	7.2
Public Renter	34	2.7	10	2.6	4	3.6	48	2.7
Rent Free	36	2.8	8	2.1	2	1.8	46	2.6
Total	1280	100	379	100	110	100	1769	100

Note: Couples who have been in one de facto relationship or a number of de facto relationships are excluded from the analysis as the data does not provide information on the number of past de facto relationships.

Source: Authors calculations from Wave 1 of HILDA

In table 12 we ask whether the marital history of older Australians differs by age group and gender. Are particular groups of older Australians more or less likely to churn in and out of marriages? In view of the findings in table 11 this is an important issue. Our analysis of marital histories employs the same age bands used by (Olsberg and Winters 2005). Those aged 50 to 59 are the 'senior baby boomers', the 60 to 74 year olds grew up during World War II, and those over 75 had experienced hardship growing up during the depression years (Olsberg et al. 2004; Olsberg and Winters 2005). These age categories allow us to examine how marital histories and the housing careers of each group differ given the contrasting economic climates during adolescence and young adulthood. There are 17% of older Australians who have remarried once or more. We find that remarriage shares are higher among the senior baby boomers compared to the older age groups. These remarriage shares are more significant than might appear at first sight since senior baby boomers are younger and have therefore had less 'time' to churn in and out of marriages. There is also some evidence of polarisation in marital histories since the proportion of senior baby boomers who have never married (5%) is higher compared to 60 to 74 year olds (4 %) and those 75 and over (3 %). The senior baby boomers are clearly less attached to their partners as compared to older age cohorts examined in table 12. There is little difference in the marital histories of males and females: 78 % of males and 81 % of females have been married once; 17 % of males and 16 % of females have married two or more times.

Table 12 - Marital History of Older Persons, 2001

Tenure	50 to 59		60 to 74	60 to 74		75 and over		
	Number	%	Number %		Number	%	Number	%
Never Married	146	5.5	46	4	26	3.4	218	4.8
Married Once	2014	75.9	920	80.9	641	82.9	3575	78.4
Married Twice	428	16.1	154	4 13.55 98 12.7		12.7	680	14.9
Married Three + times	64	2.4	17	1.5	8	1	89	1.9
Total	2652	100	1137	100	773	100	4562	100

Note: The analysis excludes individuals in an uncategorised tenure and individuals who are legally married but not living with their partner.

Source: Authors calculations from Wave 1 of HILDA

However, there is an important gender dimension to consider because among all older singles, *over 70 % are female*. Older singles that have lost a partner are also typically women. On the other hand, males account for 59% of older singles that never married.

These gender contrasts likely reflect the greater longevity of women, and their lower propensity to remarry following loss of a partner. Indeed, older single females that have lost a partner are typically widowed (69% of all such females), but older single males are more likely to be divorced (43% of all such males) than widowed (36% of all such males).

We now turn to the income position of older singles that have lost a partner. Among singles who have lost a partner, 71 % received some kind of government benefit, pension or allowance in the financial year 2000 - 2001 compared to 44% of couples, and 47 % of never married income units. The higher rates of eligibility for income support programmes (ISP) among proportion of singles that have lost a partner raises concerns about the financial welfare of this group. Since 30 percent of older singles that have lost a partner are renters (see table 10), and eligibility for ISP is a 'passport' to housing assistance, this segment of the older population of Australians will have relatively high levels of demand for housing assistance (see section 2.3 below, where this is explored further). This is particularly the case for older singles that have lost a partner due to divorce and separation.

Table 13 examines the income distribution of older Australians by marital status. There are considerable differences between singles and couples – nearly 80 % of singles who have lost a partner, and 70% of those who never married fall in the bottom 40 % of the income distribution, whereas 40% of couples are in the highest two income quintiles. Singles that have lost a partner are considerably over represented in the lowest income quintile – 78% of those in the bottom 20% of the income distribution are singles who have lost a partner. Since 72% of singles who have lost a partner are female, it is women who are most affected by the inferior income position associated with loss of partner.

Nearly nine in ten (87%) widows are represented in the lowest two income quintiles compared to 74 % and 65 % of separated and divorced singles respectively. Separated and divorced singles are on average younger than widows (average age of separated singles is 59, divorced singles 61 and widows 74). Widows are then more reliant on occupational or state pension income, and unlikely to have income from earnings. However if we standardise for age by examining the 65+ years age group, the median gross income of separated and divorced singles is \$10,700, which is lower than that of widows at \$11,000.¹³ This suggests that once we control for age, separated and divorces singles have a marginally inferior economic wellbeing as compared to widows.

. Widow Pension is expected to decline as veterans from World War II age (Costello, P. (2002).

25

Over half (54 %) of singles that have lost a partner received the Age Pension compared to 43 percent of couples and 47 percent of never married income units; 9 percent received Disability Support Pension (16 percent couples and 27 percent never married), and 8 percent the War Widow Pension. The uptake of War

Intergenerational Report 2002-03. 2002-03 Budget Paper No. 5. Canberra, Commonwealth of Australia.

¹³ Based on a sample of 140 separated and divorced income units, and 522 widows

Table 13 - Gross Income Distribution¹ (Defined by Total income unit sample) of older Australian income units by marital status, 2000-01.

	Married/ de facto	Single,	lost a partne	Single, never	Total			
	All Separa		Separated	d Divorced Widowed		legally married		
Total	N	1770	1125	137	350	638	197	3092
1st income quintile	%	5.2	61.1	51.1	50.0	69.3	51.3	28.5
(>\$0 and < \$14,849)								
2nd income quintile	%	35.8	18	23.4	15.4	18.2	18.8	28.2
(≥ \$14,849 and <\$27,342)								
3rd income quintile	%	19	12.6	10.9	21.7	8	16.8	16.6
(≥ \$27,342 and <\$46,884)								
4th income quintile	%	16.7	5.5	8.8	8.9	3.0	8.1	12.1
(≥ \$46,884 and <\$74,897)								
5th income quintile	%	23.3	2.8	5.8	4	1.6	5.1	14.7
(≥\$74,897)								
Total	%	100	100	100	100	100	100	100

Note 1: Gross income is calculated by adding together private income from the previous financial year (which is the sum of financial year gross wages and salary, financial year business income, financial year investments and financial year private transfers), and government transfer entitlements (which is the sum of pension, allowance, family benefit or supplementary benefit entitlements). Income quintiles in table 13 are defined using gross income of all income units aged over 15.

Source: Authors calculations from Wave 1 of HILDA

2.2 Housing affordability among older Australians that have lost a partner

In this sub section we examine the affordability of housing for a sample of older Australian income units using Wave 1 of the Housing, Income and Labour Dynamics Australia (HILDA) panel survey. This preliminary analysis provides background information to future analysis of the housing and financial circumstances of income units before and after the loss of a partner. This section also provides evidence directly relevant to the first research question of the proposal: do the housing circumstances and after housing cost incomes of older Australians that have lost a partner compare unfavourably with other household types. Our categories of household types are based on marital histories; the categories are coupled (married or living with a partner in a de facto relationship), single – lost a partner (separated, divorced, widowed) or never legally married. Income units that identified as married but not living with their partner were excluded from the sample.

As in section 2.2 we analyse a sample of income units where at least one person is 50+ years. The housing tenure of income units are outright owners, owner purchasers, private renters, public renters or living rent free (uncategorised or boarders in non-

private households where amount paid is unknown are excluded from the sample). Since outright owners have zero mortgage payments, and HILDA contains no information on repairs and maintenance outlays their housing affordability circumstances are not examined. Two-thirds of older Australians are outright owners, so housing affordability stress is potentially a problem among one-third of older Australians. Public housing tenants are not examined as their rents are set at 25 percent of assessable income. In principle public housing tenants are, like outright owners, not vulnerable to housing affordability stress (HAS).

We measure housing affordability using the ratio of housing costs to gross income from all sources (HAR)¹⁴. Housing costs are the mortgage repayments of home purchasers and the gross rents paid by renters. This measure is commonly referred to as the gross housing affordability ratio. The net housing affordability ratio is calculated for private renters, and is obtained by subtracting their Commonwealth Rent Assistance (CRA) entitlements from gross rent. CRA entitlements are calculated using the AHURI-3M microsimulation model of the Australian housing market. Net housing affordability ratios are the preferred measure because they reflect tenants' actual out-of-pocket housing costs. Finally, housing affordability stress (HAS) is identified by the benchmark ratio of 30%, a figure that has become a convention among Australian housing policy analysts.

2.2.1 Older private renters and Housing Affordability Stress 15

Table 14 describes the profile of older private renters (10% of all older income units) according to their income unit type (column 2), their median HAR (column 3) and the proportion of each income unit type that are found to be suffering HAS (column 4). Though singles that have lost a partner comprise 29% of the population of older Australians (see table 3), they account for a disproportionately high 51% of older private renters. The median HAR is 25%, though widows and separated singles have a slightly higher median HAR. Among all older private renters, 37% are in HAS according to the gross HAR measure; separated and divorced singles are more vulnerable with 55 and 41 %, respectively, of this group in HAS.

$$HAR = \frac{\sum \frac{Ri}{Yi}}{N}$$
 and $HAR = \frac{\sum \sum \frac{Ri}{N}}{\sum \sum \frac{Yi}{N}}$

Where HAR is housing affordability ratio, R is rent or mortgage repayments, Y is income and i = 1,2...N is income unit i. We also calculated two median measures, the median of HAR_i from the sample of income units and

$$\frac{median R_i}{median Y_i}$$

The findings reported here use the median of HAR_i . Median measures are not sensitive to extreme values that can distort means, particularly in small samples. The alternative measures corresponding to the statistical analyses reported in tables 14 - 19 can be obtained from the authors on request.

¹⁴ We experimented with the mean ratios as calculated using the formula shown below:

¹⁵ Estimates should be interpreted with caution as they are based on small sample numbers.

Table 14 - Median HAR¹ and Incidence of Housing Affordability Stress (HAS) Among Private Renters Aged 50+: before Commonwealth Rent Assistance.

Marital status	Number and % of older IUs ²		Median Gross HAR	Numbers and % of Income Units Housing Affordability Stress ³	
	N	%	%	N	%
Couples	123	42%	23	39	32%
Separated	31	10%	31	17	55%
Divorced	68	23%	25	28	41%
Widowed	52	18%	26	20	38%
Never married	22	7%	18	6	27%
Total	296	100%	25	110	37%

Note 1: HAR excludes cases where HAR is more than 100 %. Six income units have been excluded from the sample for having a HAR over 100 %. Fourteen income units with a negative or zero income have been excluded.

Note 2: Population estimates for marital categories are: couples, 317141; separated, 17017; divorced, 52719; widowed, 28790 and never married, 14472.

Note 3: HAS is defined as income units paying more than 30 % of income in rent.

Source: Authors calculations from Wave 1 of HILDA

Table 15 presents the same information as table 16, but for net affordability ratios that take CRA entitlements into account. The incidence of HAS falls to 23% of older private renters, and the differences between older singles that have lost a partner and other household types narrows. The larger impact of CRA for singles that have lost a partner is due to their greater eligibility for government income support payments that act as a 'passport' to CRA eligibility (Wood et al. 2005) for details on CRA eligibility rules). CRA cushions the housing cost burden of a high proportion (64%) of renting singles that have lost a partner. However, even after CRA is taken into account a little under 1 in 4 older singles who are divorced or separated are in HAS (see table 15). Among older couples this rises to 1 in 4.

Table 15 - Median HAR¹ and Incidence of Housing Affordability Stress (HAS) Among Private Renters Aged 50+: after Commonwealth Rent Assistance

Marital status	Number and % of older IUs ²		Median Net HAR	% of Income Units in Housing Affordability Stress ³		
	N	%	%	N	%	
Couples	123	42%	21	32	25%	
Separated	31	10%	23	8	26%	
Divorced	68	23%	21	19	28%	
Widowed	52	18%	20	5	10%	
Never married	22	7%	17	3	14%	
Total	296	100%	21	67	23%	

Note 1: See note 1, table 14

Note 2: Population estimates for marital categories are: couples, 1421601; separated, 27967; divorced, 68440; widowed 51381, and never married, 21283.

Note 3: See note 3, table 14

Source: Authors calculations from Wave 1 of HILDA

2.2.2 Home purchasers and Housing Affordability Stress 16

Table 16 has the same design as table 14, but reports findings for older home purchasers, who account for 14% of all older income units. Note that home purchasers are not eligible for CRA. The median HAR is 15%, somewhat lower than the median (net) HAR among private renters (21%). The proportion of older home purchasers in HAS is 17%, which is again lower than the 23% figure for private renters.

Divorced and separated home purchasers have a median HAR of 17 and 15 %, respectively, as compared to 13 % for couples. HAS is more likely among singles that have lost a partner, particularly divorcees 27 % of whom are in HAS. Only 12% of widows are in HAS.

Older home purchasers have median mortgage repayments of \$560 per month and a median outstanding mortgage debt of \$56,000. It could be that these are high income, two earner income units who can expect to repay these mortgages in a short period of time. It turns out that 90 % of these home purchasers are younger than 65 years of age, and can therefore expect to have a stream of earnings to help meet repayments. The median income of older home purchaser income units is \$52,300, and this compares to average incomes of \$22,900 among all older income units. There are only 9 % of older home purchasers in the lowest income quintile of the Australian population. This suggests that older home purchasers are typically drawn from middle to high income groups; *if their incomes are secure* then mortgages will eventually be paid off and HAS levels will ease over time.

Table 16 - Median HARs and Incidence of Housing Affordability Stress Among Home Purchasers +50.

Marital status	Number and % of older IUs ¹		Median HAR	% of Incon Affordability Stre	
	N	%	%	N	%
Couples	266	70%	13	37	14%
Separated	18	5%	15	3	17%
Divorced	55	14%	17	15	27%
Widowed	26	7%	11	3	12%
Never married	15	4%	21	5	33%
Total	380	100%	15	63	17%

Note 1: Population estimates for marital categories are: couples, 317141; separated, 17017; divorced, 52719; widowed, 28790; and never married, 14472.

Note 2: HAS is defined as income units paying more than 30 % of income in housing costs.

Source: Authors calculations from Wave 1 of HILDA

Table 17 combines home purchasers and private renters to offer a profile of the housing affordability position of all older Australians that are potentially vulnerable to HAS. The median HAR is 18%, and 19% of older Australian purchasers and private renters are in HAS. Among singles that have lost a partner, divorcees are more vulnerable to HAS and widows are the least vulnerable.

29

¹⁶ Estimates should be interpreted with caution as they are based on small sample numbers. This is particularly the case for never married older persons.

Table 17 - Median HARs and Incidence of Housing Affordability Stress Among Home Purchasers and Private Renters: after Commonwealth Rent Assistance

Marital status	N/Percentage of older IUs		Median HAR	Percentage of Income Affordability Stress ²	e Units in Housing
	N	%		N	%
Couples	389	58%	15	69	18%
Separated	49	7%	22	11	22%
Divorced	123	18%	20	34	28%
Widowed	78	12%	19	7	9%
Never	37	5%	19	8	22%
Total	676	100%	18	129	19%

Note 1: Population estimates for marital categories are: couples, 459301; separated, 44983; divorced, 121159; widowed, 80171; and never married, 35754.

Note 2: HAS is defined as income units paying more than 30 % of income in housing costs.

Source: Authors calculations from Wave 1 of HILDA

We need to keep in mind that many if not most of these older Australians are past the peak of their life cycle income profile, and can expect lower real incomes later in their lives. For older home purchasers that are able to repay mortgages, declines in real income are not so problematic. For older private renters, the prospect of lower real incomes is particularly worrying if real rents remain constant or increase. We explore this issue in table 18 where the HAR and incidence of HAS among older private renters is presented across the Olsberg and Winter age bands originally employed in table 10 above.

Table 18 - Median HARs and Incidence of Housing Affordability Stress Among Private Renters: Before Commonwealth Rent Assistance

Age Bands	Number and % of older IUs ¹		Median Gross HAR	% of Income Units in Housing Affordability Stress ¹	
	N	%	%	N	%
50 to 59	154	53	24	53	34
60 to 74	101	34	28	46	45
75 and over	37	13	26	11	30
Total	292	100	25	110	38

Note 1: HAS is defined as income units paying more than 30 % of income in housing costs.

Source: Authors calculations from Wave 1 of HILDA

Table 19 - Median HARs and Incidence of Housing Affordability Stress Among Private Renters: After Commonwealth Rent Assistance

Age Bands ¹	Number and % of older IUs ²		Median Net HAR	% of Incor	me Units in Housing y Stress ¹
	N	%	%	N	%
50 to 59	154	53	21	39	25
60 to 74	101	34	21	25	25
75 and over	37	13	20	2	5
Total	292	100	21	66	23

Note 1: The age is used to clarify income units by age band. In the case of couples, the reference person in couples is in most instances a male. There are 123 coupled income units that are older private renters; there are 4 cases where the reference person in a couple is younger than 50, and have therefore been excluded from the analysis.

Note 2: HAS is defined as income units paying more than 30 % of income in housing costs.

Source: Authors calculations from Wave 1 of HILDA

Unfortunately sample numbers in the age band 75 and over are too small to base any firm conclusions upon. However there are better sample numbers in the other two age bands. Senior baby boomers (50 to 59) have lower gross HAR and lower rates of HAS as compared to the older 60 to 74 years private renters. This would indicate that progression into retirement is associated with growing housing affordability problems. However, on taking CRA into account the adverse housing affordability circumstances of 60 to 74 years group are mitigated. The higher CRA eligibility rates among 60-74 year olds help lower their housing costs as they enter retirement and incomes fall.

2.3 Summary

In assessing the likely future demand for housing assistance from older single persons, the demographic background is a critical factor. Our examination of relevant demographic trends uncovered the following key points:

- The Australian population is ageing and this process will accelerate as the baby boomer generation enters retirement.
- Among older (50+years) Australians the share of persons that are divorced, separated or widowed has increased from 24% in 1981 to 29% of all older Australians in 2001. Older single persons who have lost a partner are then a rapidly growing segment of the population. There is an important gender dimension, as 72% of older single persons that have lost a partner are female.
- Death of a partner is becoming a relatively less important cause of household dissolution among older Australians. Divorce and separation are becoming correspondingly more important. The number of older Australians that are separated or divorced increased from 184,000 in 1981 to 705,000 in 2001.
- Marriage rates (number of marriages per 1000 adults) have been declining in Australia. In 1983 the rate was 7.5, and this declined to 5.4 in 2003. The median age of marriage has increased. If trends continue we will witness an increasing number and incidence of never married older Australians; this is not yet apparent in the current cohort of older Australians.
- Divorce rates (number of divorces per 1000 adults) are at historically high levels, but have remained stable over the period 1983 to 2003. The median age at which people are divorcing is increasing, and there is a corresponding increase in divorce rates among older Australians.
- The propensity of divorcees to remarry is declining. There is <u>net</u> growth in the number of single person households formed as a result of divorce because only around 70% of divorcees remarry.

We used the HILDA data base and AHURI's 3-M Microsimulation Model to profile the housing tenure and housing affordability position of older singles who have lost a partner. The following key findings warrant emphasis:

- Outright home ownership is a very important characteristic of older Australians; two-thirds of older Australians are outright owners. However, a sizeable minority of older Australians (16%) are in private or public rental tenures.
- Older singles that have lost a partner are much more likely to rent one-quarter
 of this group rent from a private or pubic landlord. The divorced and eperated
 are particularly reliant on rental housing.
- Marriage dissolution might have adverse impacts on housing outcomes that are not completely reversed on remarriage. We find that divorcees that repartner have lower rates of outright ownership compared to the continuously married.
- Older singles that have lost a partner have a high enrolment in income support programmes, with 71% receiving some kind of government benefit, pension of allowance. Older couples have a much lower rate of entitlement with 44% receiving an income support payment.
- Older Australians that have lost a partner are much more likely to have low incomes. Of this group 68% have incomes in the bottom 40% of the income distribution; 38% of older couples are found in the bottom 40%. Older widows are particularly vulnerable to low income outcomes, but have higher rates of home ownership (including outright ownership)

- Among private rental tenants and before taking CRA into account, the incidence
 of HAS is relatively higher among older singles that have lost a partner. But
 because this group are more likely to be eligible for CRA, their relatively
 disadvantaged position is more or less corrected once CRA is taken into
 account.
- Older Australians that are home purchasers are less likely to suffer HAS, but the incidence of HAS is again relatively high among older singles that have lost a partner and they are ineligible for CRA.

3 NEXT STAGES IN THE RESEARCH

The research team will in the remainder of this project proceed with two strands of research. One strand is quantitative and employs various secondary data sets to measure the impact of loss of a partner on the housing and financial circumstances of those affected. Two data sets will be instrumental in the analysis – the Household Income and Labour Dynamics in Australia Survey (HILDA) and the Department of Family and Community Services Longitudinal Dataset of benefit recipients. The way in which these datasets will be used and the research questions addressed is explained in subsections 3.1 and 3.2 below.

A second component of the quantitative analysis will be concerned with projecting future demand for public housing from that group of the adult population that experience loss of a partner. The main secondary data set that will be used is the administrative records maintained by Homewest in Western Australia. We describe in Section 3.3 how these records will be used for the purpose of projecting future demand for public housing.

The remaining strand of research is qualitative and will conduct an in depth examination of the housing implications of the loss of a partner for older people. There will be 60 in depth interviews the question themes that we plan to explore are outlined in section 4 below.

3.1 Loss of partner and the consequences for housing circumstances: a longitudinal analysis using HILDA

3.1.1 Key research questions

Waves 1-3 of the Household Income and Labour Dynamics Survey will be a critical data source because the longitudinal nature of the survey data permits exploration of the pathways preceding and following loss of a partner. In this survey there are 1,872 older (+ 50 years) persons that have lost a partner at some point in their life course. In the 24 months covered by waves 1-3 there are 127 older persons that have lost a partner. These persons and their confidentialised unit records will comprise key samples for analysis of the following key research questions.

- Do the housing circumstances and after-housing cost incomes of older Australians that have lost a partner compare unfavourably with other household types?
- How are the housing and financial circumstances of older persons affected by loss of a partner?
- Is the probability of losing a partner higher among persons that are 50 + years of age, private rental tenants or income support recipients?

These research questions are central to the major issue addressed in this project – does household dissolution due to divorce, separation and bereavement lead to increasing demand for housing assistance from older persons?

3.1.2 Approach

Section 2 of this positioning paper has presented some initial findings on how the housing affordability position of older persons that have lost a partner compare with the rest of the older population of Australians. This analysis was based on the retrospective marital histories of a cross section sample of wave 1 respondents. It is a 'static' comparison of persons that have lost a partner where these persons are at different points of the pathways of adjustment to this adverse event.

The statistical analysis confirms that loss of a partner is associated with inferior home ownership outcomes and a higher incidence of housing affordability stress among older

Australians that suffer loss of a partner. An appealing feature of HILDA is the panel nature of the survey. It follows the same sample of persons tracking this panel over successive waves and allowing researchers to move beyond static comparisons. It allows dynamic analyses of adjustment pathways with respect to events such as household dissolution, the focus of this project.

As is pointed out in our research proposal (see page 6) there are 127 older persons that have lost a partner between wave 1 and wave 3. This sample offers us the opportunity of tracking the financial circumstances and housing careers before and after household dissolution. Of particular interest are the immediate consequences for and adjustments made by persons that have lost a partner, and their housing career adjustments will be our focus. A key task will be documenting the changing demand for housing assistance and the AHURI-3M microsimulation model will be used to identify those who become eligible for Commonwealth Rent Assistance and/or public housing in the aftermath of divorce, separation and bereavement. Eligibility for public housing will be determined by the application of state housing authority income eligibility limits to measures of assessable income.

Though older persons are the group of most relevance to the project, it is to be noted that our demographic analysis revealed that divorce and separation typically occur before Australians turn 50. Restricting the analysis to older Australians may prevent identification of younger divorcees that suffer permanent scarring to their housing careers, resulting in a demand for housing assistance that lasts into old age. We therefore plan to conduct analysis on the housing careers of all coupled individuals (legally married or de facto) in wave 1 of HILDA and experience loss of a partner due to divorce, separation or bereavement between waves 1 and 3.

Loss of a partner is more likely to result in a demand for housing assistance if low income households are more vulnerable to loss of a partner. We will draw on the work (Grossbard - Shechtman 1993; Grossbard - Shechtman 2003; Lehrer 2003) to design model specifications that allow us to measure variation in the probability of household dissolution across different sub-groups of the population. Our approach will be to take the sample of all Australian households and use their retrospective marital histories, as recorded in HILDA, to identify those who are divorced, separated or widowed and their age at loss of partner. The probability of loss of partner and the relative importance of variables determining variation in probabilities will be estimated in two models, one specification for widows the other for those divorced and separated. This is a sensible approach because the variables linked to the chances of divorce and separation is likely to be different from those impacting loss of a partner due to bereavement. Even when the same variable is included in both models the direction and strength of the causal link could differ. Take age for example. Household dissolution due to bereavement is more likely the older the person, but the chances of being a divorcee or separated may peak in middle age as repartnering reduces their incidence in older age groups. Widows are much less likely to repartner and so the link with age is more straightforward.

The explanatory variables in these models will include housing tenure variables that allow us to explore whether particular groups eligible for housing assistance are more vulnerable to household dissolution. The findings from this modelling exercise will therefore help to inform policy makers on how demographic events such as household dissolution impact on the demand for housing assistance. The expected outcomes and policy significance of the research will be explained in section four below.

3.2 The Department of Family and Community Services' Longitudinal Dataset of benefit recipients

In 1999, FaCS decided to make available to external researchers a randomly selected, confidentialised sample of records from the Department's database used for the administration of income support payments. Known as the FaCS Longitudinal Dataset (LDS), the data comprised of the fortnightly administrative records of a 1 % sample of benefit recipients nationwide from 1995. The beneficiaries are identified by a unique customer number, allowing identification of a record for every fortnight an individual selected for the sample is in receipt of benefits. In appropriate circumstances, partner records are also included. The set of records for each individual can be collated to create a longitudinal picture of each individual's utilisation of the benefit system and their relevant circumstances.

Unfortunately FaCS have recently decided to review the provisions of access to the LDS by external researchers. For the analysis planned for this project, this means the research team cannot be certain as to the exact nature of the data which will become available, including the period covered by the fortnightly records, the variables to be included or the sample size. Previous releases of the data included all fortnightly records from June 1995, and it is anticipated that what data is made available would also commence from this time, meaning the sample would cover income support recipients over a period of around 10 years. The LDS covers such a large number of individuals that sample sizes are rarely a binding constraint when working with these data.

The LDS has both limitations and strengths. As the data are collected with the primary purpose of administering income support payments a limited and very focussed set of information is collected for each individual, particularly when compared to broad ranging surveys such as HILDA. There is, for example, no usable information on level of education, previous occupation or work experience, or other labour market and socio-economic variables known to be important correlates with many outcomes in life. However, the data that is collected is done so with a high degree of accuracy, since it is used to determine individuals' eligibility for income support and the amount of that payment.

For the purposes of this project, the principal variables available include:

- Age and gender.
- Marital status either couple or single.
- Type and amount of income support payment including unemployment benefits (Newstart), age pension, widow and bereavement allowances.
- Home ownership status including non-home owner; home owner (purchaser); home owner (outright); owner or non-owner living in government-funded aged care facilities; plus a range of values for more specific states.
- Commonwealth Rent Assistance eligibility and amount.
- Rent type paid including private rent, government rent, maintenance fees paid to a nursing home or retirement village.
- Income earned and unearned.
- Assets value of a large range of assets required for asset testing to determine payment entitlements where applicable.

It is important to keep in mind that the sample is not representative of the Australian population as a whole, but only of those who were in receipt of income support payments in the period covered by the data. Therefore it cannot be used to make inferences about the population as a whole. However it can be used to make

inferences about the impact of loss of a partner and other factors within the group of income support recipients. As this group includes recipients of the age pension, it still represents a significant proportion of persons aged 50 years and over, and an increasing proportion of older groups. It is also not possible to identify whether or not a single person lost a partner if this occurred prior to becoming an income support recipient or if the separation occurred prior to the first date of the period covered by the LDS, unless the person is in receipt of a widow or bereavement related payment. People may have been single all their lives, although as detailed above such people are a minority for the 50+ age group. Separations that occur during a spell of income support receipt or between spells can be identified through transitions from 'couple' to 'single' marital status. In some cases it will be possible to identify whether this separation occurs due to the death of a partner due to payment types.

The exact methodological approaches taken in analysing the LDS will depend on how the data is made available, as outlined above, and initial testing once access is achieved. Three broad approaches have been identified at this stage:

Descriptive comparisons – this approach will essentially treat the data as a series of cross sections to compare the circumstances of those with and without partners. Comparisons will be made of housing ownership/rental status, income and assets for like people (eg. by age and gender) conditional on whether their contemporaneous marital status is 'couple' or 'single' (single treated as a broad proxy for those who lost a partner). Comparisons will then be made for subsets of the 'single' persons restricted to persons who are positively identified to have experienced a separation and for whom that separation can be identified as having occurred through bereavement.

Transition analysis – the longitudinal nature of the data will be exploited by observing changes in housing ownership/rental status over selected periods (eg. after one year, two years, five years) following the observed loss of a partner and compared to those of persons continuously partnered and continuously single. Account will be taken of additional factors by estimating a multivariate probit models to estimate the likelihood of selected transitions between housing states, such as from non-ownership to ownership; ownership to non-ownership, into government rental accommodation and into government-funded aged care facilities. Where possible, these models will also be estimated as panel models to take account of fixed individual effects. However, the robustness of this approach may be compromised if the number of "censored" observations is too great.¹⁷ For periods in which individuals are not receiving income support and have no fortnightly LDS record, their outcomes cannot be determined and the observations would have to be omitted. This raises the possibility of bias in the results if leaving the data is correlated with the explanatory or outcome variables. Longitudinal probit models will be estimated using STATA's 'xtprobit' routine with both random and fixed effects.

Hazard analysis – this approach will be used to complement the transition analysis. The two approaches are conceptually equivalent except that the hazard analysis is done in a framework of 'continuous time' rather than for selected elapsed periods. The hazard rates for transitions identified above will be estimated, where the unit of observation is a spell of income support receipt and the time unit is the individual's age in fortnights. Thus, for example, the probability of moving from home-ownership to non-ownership would be estimated by fortnight from age 50 onwards. Non-parametric hazards will be presented (ie. the actual hazard rates observed in the data) initially for couples and singles. An appropriate functional form will then be selected to allow estimation of the impact of covariates on the hazard rate, such as gender and age at the beginning of the spell, with loss of a partner included as a time varying covariate.

_

¹⁷ Panel data is collected over a finite time frame. Those individuals who have yet to make a transition from one statue to another represent observations that are (right) censored.

These approaches will be used to provide evidence on Key Research Question 1 (Do the housing circumstances of older Australians that have lost a partner compare unfavourably with other household types), Research Question 2 (How are the housing and financial circumstances of older persons affected by loss of a partner?), and Research Question 4 (Do older persons experiencing partner loss enter public housing?). Research Question 3 (Is the probability of losing a partner higher for Older persons (+50) and private rental tenants) will also be addressed using the methods set out above, but with marital status as the outcome and housing status included as conditioning variables.

3.3 Homeswest tenancy records and projecting the demand for public housing

A dataset compiled from Homeswest's waitlist and tenancy records will be combined with ABS Census data and ABS population projections to provide evidence on Research Question 4: Do older persons experiencing partner loss seek to enter public housing? Are they successful in entering public housing? and Research Question 5: How large is the group of older persons that have lost a partner likely to be in the next 10-20 years? What implications might this have for public housing demand? In separate analyses, the 1% sample file from the 2001 Census will be used to investigate Research Question 6: What are the implications of loss of partner for residential location of older persons?

The approach is to first use the Homeswest data to identify the number of persons from groups of interest seeking to enter public housing, and the time taken to get into public housing after first applying. The main group of interest is single persons aged 50 years and older. The number of people in these groups seeking to enter public housing will then be compared to their respective total populations within Western Australia, as taken from the Census, in order to approximate the *propensity* of each group to seek to enter public housing in 2001. Assuming this propensity and the proportions of single persons by age will remain constant into the future, the ABS Population Projections by age and gender can then be used to project demand for public housing by older, single persons to 2025. Estimates for other states will be made using population estimates for these states and applying the Western Australian propensities to enter public housing.

3.3.1 Data Description

Homeswest tenancy records for applicants entering the waitlist after January 1999 that were successful in entering public housing by November 2005 are examined. Applicants that either withdrew their application, or remain on the waitlist are not included in these data. We are advised that data covering applicants that did not secure public housing will be made available for later analysis. Applicants that entered public housing more than once are considered only on their first entry.

Each record includes an application date and occupancy date that are used to determine the length of time before an application is successful. An application is made by households and personal data is recorded for all members, including date of birth, relationship status, and gender. The relationship code (RLCD) identifies the applicant as being in one of 17 categories, which include 'single adult' and 'sole parent'. However, we note that the data were updated and overwritten by Homeswest as the applicant's relationship circumstances change subsequent to the initial application. Further, we note that a number of applicants classed as 'single adults' are accompanied by a minor. Thus, in order to correctly identify the 'single adult' and 'sole parent' groups at the time of the initial application, more detailed analysis is required.

The ABS 2001 Census 1% Unit Record File records the number of households by household type in WA. ABS Population Projections for WA are available by age and gender for years 2006 to 2025. The population projections are used to forecast changes in these demographic groups based on assumptions about future rates of fertility, mortality, internal migration, and net overseas migration. Series A assumes high levels of fertility, net migration, and life expectancy with respect to existing levels. Series B assumes a continuation of current levels of fertility, migration and life expectancy while Series C assumes a decline in each of these levels. Using appropriate weights, the data from the 1% Census file will be used to generate population estimates of the number of older single person and sole parent households for WA in 2001. Holding these proportions constant within age and gender categories, the ABS projection will be used to estimate the number of such households containing older, single persons out to 2025.

3.3.2 Preliminary Results - Do older persons experiencing partner loss seek to enter public housing? Are they successful in entering public housing

The 'waitlist time' – the time on the Homeswest waitlist – is calculated for each record as the period from when the application is first made to when that tenant enters Homeswest housing. While the number of persons within the relevant population entering the waitlist is used as the indicator of whether or not persons seek to enter public housing, the waitlist time is used an indictor of their success in entering public housing. Where an application has been made and the applicants remain on the waitlist at the time the records were extracted, it is not possible to calculate the waitlist time. Such observations are known as 'censored' observations. For censored observations we can observe the elapsed time on the waitlist, but not the final outcome with respect to their 'waitlist time' before entering public housing.

The distribution of the time that an applicant is on the Homeswest waitlist before securing public housing (waitlist time) is positively skewed, with 10% of applicants taking longer than 3 years and 5% taking longer than 4 years. Accordingly, there is minimal censorship of the data for the group of applicants entering the waitlist in 2001 through the exclusion of those that obtained housing after November 2005. Censorship is more pronounced in the earliest and latest time samples in the dataset, compromising time-series comparison. One method of reducing the censorship problem and permitting comparison over time, is to define the 1999 applicants as those entering the waitlist in 1999, and to define the 2005 applicants as those that entered public housing in 2005 (but may have entered the waitlist some years earlier). That is, for those who entered the waitlist in 1999, it is possible to observe how long it took them to later enter housing, with minimal censoring. For those who entered housing in 2005, it is possible to observe how long they had been on the waitlist prior to entering housing, with minimal censoring. Thus estimates of the distribution of time spent on the wait list prior to entering housing for two different periods can be obtained.

The data contains applicants' dates of birth, the date they enter the waitlist and the date they enter public housing. It is therefore possible to calculate the elapsed time spent on the waitlist prior to entering housing by age of the applicant. The dataset is filtered to select records in the required interval (e.g. joined waitlist in 1999), and applicants are grouped by age, relationship code, and gender. A count of applicants and the average waitlist time are then calculated for each group as shown in the Tables 20 (a) to (d)

Single adults and sole parents are about three times more prevalent than partnered applicants in the over 50 group of applicants seeking to enter public housing. Females are 20% more common than males in the single adult category, and three times more common in the sole parent category. Generally, for single adults over 50, the average

time on the waitlist (244 days) is fairly stable across age groups with a general decrease in waitlist time with age. Males obtain housing fractionally sooner. For sole parents, the average waitlist time is longer (likely reflecting the availability of suitable accommodation) at 314 days, but with males taking 54 days longer (on average) to place.

Persons identified as sole parents that are over the age of 70 are most likely to be living with adult children that are not dependants. As the circumstances of parents with dependant children differ from those living with adult children, subdivision of this category is appropriate.

It should be noted that the figures for applicants with partners (with and without children) refer to individual adult applicants, not households. Accordingly, it would appear that there are approximately six times more single adult and sole parent households in the over 50 age group than there are households with partners.

Table 20: Homeswest Tenancy Records – Number of Applicant and Average Waitlist Times for Selected Groups

a. Entered Waitlist 1999

Single	Adult				Sole I	Parent			Partn	er		
	Male		Female	!	Male		Fema	le	Male		Fema	le
Age	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait
0-50	968	394	754	416	326	348	250	433	615	352	192	332
50-55	141	334	192	416	13	640	45	305	65	412	60	310
55-60	126	206	156	281	11	562	26	315	68	412	63	366
60-65	123	320	148	353	3	410	14	488	72	475	55	536
65-70	110	246	138	320	6	130	9	260	48	389	38	264
70-75	64	258	117	251	3	461	6	172	56	338	31	371
75-80	37	177	68	248	1	474	1	1	22	282	12	326
80-85	17	148	36	185			1	4	12	240	4	184
85+	7	59	23	244			1	303	8	142	1	122

b. Entered Waitlist 2001

Single	Adult				Sole I	Parent			Partne	r	_	
	Male		Female		Male		Fema	le	Male		Fema	ile
Age	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait
0-50	124	274	888	279	339	337	257	339	768	233	188	258
50-55	165	235	176	274	16	337	56	316	59	232	53	271
55-60	140	205	133	265	4	588	28	368	62	314	45	313
60-65	132	235	160	259	6	248	13	398	76	366	61	371
65-70	112	296	148	285	2	637	12	101	63	383	44	476
70-75	91	197	108	248	2	439	3	75	39	451	27	333
75-80	56	206	68	208	2	2	4	110	24	269	11	409
80-85	24	275	49	166					11	227	10	289
85+	11	127	29	199					2	30	2	201

c. Entered Housing 2005

Single	Adult				Sole F	Parent			Partner			
	Male		Femal	e	Male		Fema	le	Male	÷.	Female)
Age	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait
0-50	1071	385	783	386	294	453	228	456	563	365	186	346
50-55	165	306	153	382	29	538	56	361	53	162	47	428
55-60	131	285	151	275	17	531	35	284	52	384	49	464
60-65	119	250	127	347	7	459	22	389	52	491	34	389
65-70	102	221	113	292	2	247	9	106	40	383	30	609
70-75	82	188	99	282	2	427	7	441	43	508	26	369
75-80	38	193	82	272	3	423	3	111	25	437	13	436
80-85	14	226	37	140			1	74	8	670	3	587
85+	8	271	28	154					2	872		

d. Averages for Over 50 Age Group

	Single	Adult			Sole F	Parent			Partne	er		
	Male		Femal	le	Male		Fema	le	Male		Fema	le
Age	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait	No.	Wait
1999	625	265	878	317	37	497	103	315	351	390	264	369
2001	731	231	871	255	32	356	116	302	336	332	253	351
2005	659	255	790	300	60	507	133	324	275	397	202	452

Waitlist times are remarkably stable for all categories between 1999 and 2005. The marginal reduction of the average waitlist times for 2001 may be attributable to censorship of the small group of applicants with the longest waitlist times. There is a reduction in the number of single women, and partnered people seeking to enter public housing, with a numerically small increase in sole parents of both sexes. The robustness of our findings with respect to mean waitlist times will be checked using survival analysis techniques which allow estimation of the mean duration of completed spells taking account of censored observations.

3.3.3 Preliminary findings – How large is the group of older persons that have lost a partner likely to be in the next 10-20 years? What implications might this have for public housing demand?

The proportions of the population in the 2001 Census that fall into the 'single adult' and 'single parent' categories (further subdivided by gender and age group) are applied to the three WA population projections. The estimated number of Western Australians in these categories with sensitivity analysis for the years 2010, 2015, 2020 and 2025 is thus determined.

Using the number of persons that entered the waitlist for public housing in 2001 in the 'single adult' and 'single parent' categories for each age group (see Table 19(b)) the proportion that these comprise of the total population in the same demographic category in 2001 is computed. These proportions are assumed to persist and are applied to the ABS population estimates for the same category to project the demand for public housing. Tables 21 and 22 show preliminary estimates of future demand for public housing by age group for 'single adults' and 'sole parents' are for Western Australia.

Table 21: Single Adult (Based on ABS Series B Population Projection)

Age	Male				Female			
	2010	2015	2020	2025	2010	2015	2020	2025
50-54	195	210	212	226	216	231	234	247
55-59	200	218	235	238	202	221	237	240
60-64	215	232	255	274	260	294	323	346
65-69	169	227	245	270	213	289	326	358
70-74	112	140	188	204	72	87	118	133
75-79	73	92	116	157	82	99	121	164
80-84	41	46	59	76	69	75	91	113
85+	18	25	31	40	43	52	61	73

Table 22: Single Parent Projection of Demand for Public Housing (Based on ABS Series B Population Projection)

Age	Male				Female) <u> </u>		
	2010	2015	2020	2025	2010	2015	2020	2025
50-54	19	20	21	22	64	68	69	73
55-59	6	6	7	7	68	75	80	81
60-64	10	11	12	12	97	110	121	130
65-69	3	4	4	5	62	84	95	104
70-74	2	3	4	4	18	22	29	33
75-79	3	3	4	6	13	16	20	27
80-84	0	0	0	0	14	15	19	23
85+	0	0	0	0	3	4	4	5

Demand for all categories in all age groups is projected to increase over the period to 2025 given a continuation of current levels of fertility, migration and life expectancy. Within each category, similar numerical increases are projected for each age group, hence the largest percentage increases are in the older age groups due to population ageing effects. Noticeably, projections indicate that single parent demand for public housing for females will increase markedly over the period of interest.

3.4 What are the implications of loss of partner for residential location of older persons?

3.4.1 Data Description

Access to the ABS Census 1% sample file has been applied for, but has not at this stage been acquired.

The ABS Census 1% sample files (2001) contain data on individuals which include their age, marital status (e.g. divorced, widowed) and housing tenure (e.g. owner, private or public renter), dwelling type and other variables available in the Census. Geographic location is also available for the respondents' usual place of residence in 2001, their usual place of residence one year earlier and their usual place of residence 5 years earlier. The geographical data is available at a number of levels of classification, including statistical local area (essentially loca government area) which would allow us to identify, for example, whether the area of usual residence was inner metropolitan, outer metropolitan or regional.

3.4.2 Methodology

Older persons (50+) that have lost a partner through divorce or death at some stage in their life and who have not subsequently re-married will be identified. The location of

their residence relative to that reported one year earlier and five years earlier will be used to explore patterns of changes in residential location. This will be explored, first, through descriptive comparisons of the gross mobility of this group relative to other older persons and of patterns in the directions of those relocations. Second, multivariate logit models will be estimated within the 50 and older population of the likelihood of a person changing residential location, conditional upon age, whether or not the person has lost a partner and other mediating variables such as income, housing tenure, gender and labour force status. The likelihood of relocation between country and city areas, and from inner to outer suburban areas which may have increased or reduced services amenities will be examined.

A limitation of this analysis is that while geographic location of usual residence is known for three time periods (2001, 2000 and 1996), only current marital status is known. The implicit treatment of (current) martial status as a constant is of course most problematic for the analysis of mobility over the five year interval. For some, relocation may have occurred when they were in a different marital status to that observed in the 2001 Census, or as a result of a change in marital status. This will certainly reduce the precision of estimates, and opens the possibility of bias, although there seems no a priori reason to expect the estimate relating to the impact of having lost a partner should be biased.

3.4.3 Preliminary Results

There are no preliminary results at this stage.

4 QUALITATIVE RESEARCH AND THE SURVEY INSTRUMENT

4.1 Qualitative research: Aims

As is evident from section one's review of the literature indicates that much of the research into the housing pathways and needs of older people is generally conducted quantitatively. Studies have generally focused either on particular populations, housing tenure or a particular housing type. Some qualitative studies have provided broad based overviews of a range of issues relating to housing and older people (for example one recent study is (Olsberg and Winters 2005).

There are, however, gaps in knowledge relating to the implications of the loss of a partner on older people (AHURI 2004); (Jones et al. 2004), in particular to their housing needs, issues relating to tenure and affordability and the impact of social networks on housing decisions. The qualitative research being undertaken in this project attempts to fill in the research gap.

The qualitative interviews will an in-depth examination of the impact of loss of a partner (occasioned through divorce, separation or death) on the housing pathways of people who are fifty years of age or older. It will investigate the pathways of the older people in different tenure types (including homeowners, those purchasing a property and those in rental property). The interviews will allow us to capture the emotional impact of the loss of a partner and will also enable us to document the housing pathways including weather the respondent moved to alternative accommodation. The interview schedule will elicit information about income, household composition and conditions, health, social and family networks and changes to these following the loss of a partner and the impact of these factors on housing decisions and choices. Emphasis will be placed on understanding causal and personal factors derived from the stories told by participants. To this effect, the qualitative research will complement the quantitative research.

4.1.1 Conducting the research and interview instruments: How

There will be sixty in-depth interviews with people who have lost a partner in the last three years and who are living in the Cities of Yarra, Brimbank and Stonnington. Approximately twenty of these will be people from culturally and linguistically diverse backgrounds. Respondents will be identified with the assistance of local councils who will provide RMIT researchers with a list of elderly citizens groups. RMIT researchers will attend these groups and explain the project to the group in detail. Those who are willing to participate in the interviews will be asked to complete a screening form which records basic information relating to the circumstances in which the partner was lost, the nature of the relationship (married, defacto etc.), housing, income, employment and contact details. The in-depth interviews will be conducted on a one to one basis and will be anonymous and confidential. The in-depth interviews will be tape recorded and selectively transcribed. The interviewer will be required to complete a section at the end of the interview regarding his/her observations, outlining the most striking /important issues raised during the interview and recommending either full or selective transcription.

The interviews will be structured around key themes including the nature of the relationship, the emotional and financial impact of the loss of partner and ability to cope with looking after the property after the loss of a partner. The interviews will cover the importance of family and social networks on housing choices and the impact of all of these factors upon housing decisions. The research has been approved by RMIT University's Ethics Committee. Consent forms will be obtained from all participants prior to the interviews. Respondents will be informed of processes to protect the anonymity of their responses and the measures taken to protect the security of the data.

4.1.2 Benefits of the qualitative research: Output

The use of semi-structured qualitative interviews will allow older people to express their concerns and views in their own words and without being dependent on literacy skills. This is likely to provide good quality information on people's attitudes and expectations (Olsberg and Winters 2005). By gathering information through an in-depth interview process, the qualitative interviews will:

- Complement the quantitative analysis;
- Fill in the existing research gap on the limited knowledge of the needs, values and preferences of older people who have lost a partner in three different localities and those from a culturally and linguistically diverse backgrounds;
- Allow us to explore the intersection of various dimensions regarding location, affordability, social networks, sentimental, personal and emotional values and the impact of these on housing decisions/choices;
- Allow an understanding of consumer preferences for various housing types;
- Allow for an examination and comparison of the diverse housing needs and preferences of different groups of older people: namely those who have lost a partner through death, divorce or separation
- Provide an understanding of what older people think about and want from their housing.
- The adoption of a narrative analysis will allow us to adopt an interpretative framework of the personal stories which will be invaluable in providing predictive indications of people's attitudes and values (Olsberg and Winters 2005);
- Add to our understanding of the values, issues and concerns of these people in specific tenure types.

4.1.3 Question themes

Section A of the interview schedule concerns the nature of the relationship and includes questions pertaining to the nature of the relationship, its length, whether there are any children of the relationship and where the children live. Section B concerns circumstances surrounding the loss of the partner and its emotional impact on the interviewee: The questions in sections A and B are useful to obtain background information about the respondent, the nature of the relationship and the circumstances of the loss of partner.

The first part of Section C relates to changes to life after the loss of partner. Questions include:

- How the parties had planned their finances for retirement (eg super, self funding, savings);
- The respondent's current income source;
- How much the respondent's income and assets changed since loss of partner;
- How the respondent expected to pay for future needs (including whether there
 was an expectation of using up assets whilst alive).

The responses to these questions will probably vary depending on whether the loss of partner occurred through divorce, separation or widowhood. The general literature on older people provides useful information in relation to some of these questions (it should be noted that only some of the general issues which have been identified will be considered and an exhaustive listing of the issues will not be undertaken in the positioning paper):

Divorce and/or separation dramatically impacts on wealth accumulation. Usually, the matrimonial home is sold and the proceeds divided between the parties. Those who divorced in the last ten years and continue to remain single have about half the wealth of those who have re-partnered. In addition to lower wealth, those who remain single are also worse off in terms of having lower incomes and more debt. In addition, their superannuation is also considerably lower than the average population. As a consequence, NATSEM reports that retirement for this group does not look good (AMP - NATSEM et al. 2005)

There are differences in the asset composition of divorced single males and females. Generally upon divorce, the house or part of the proceeds of its sale has been traditionally allocated to women whilst the superannuation has remained with the male (AMP - NATSEM 2002).

In the case where one partner dies, the surviving partner usually remains in the matrimonial home. The remaining spouse may or may not decide to sell the matrimonial home and downsize to a smaller property. (Beer et al. 2006) highlight that resort to housing wealth usually only becomes an option when non housing wealth has been consumed and there is onset of illness or other important precipitating factors. (Beer et al. 2006) identify various factors including attachment to the marital home, location and desire to pass on wealth to children as some key factors influencing older people's decision to remain in the marital home (Beer et al. 2006).

For couples who are in receipt of income support, the income will usually be reduced due to the partner's loss (Beer et al. 2006).

Several writers have highlighted important issues relating to superannuation:

- Whilst it may initially appear that older people will be better off with a superannuation system, such a system was introduced rather late in their working lives and it is only those who have had strong attachment to the workforce since the 1990s that will gain substantial benefit from their superannuation during their retirement (Beer et al. 2006).
- Women are particularly disadvantaged in regard to superannuation as older women in particular have either not participated in the workforce or have not worked on a full time basis, being employed in casual or part-time work due to family or children related commitments (Olsberg 2005).
- Birrell & Healy point out that a sizeable minority of baby boomers are either earning very low incomes or are dependent on welfare as they had to depend on volatile employment options (Birrell and Healy 2005). Many were affected by structural changes which followed on from the post war economic boom period (Birrell and Healy 2005). Furthermore, in the baby boomer population, there are high numbers immigrants from culturally and linguistically diverse backgrounds who were severely effected by the restructuring of Australia's manufacturing sector in the post 1970s and continued to be adversely effected by virtue of their lack of proficiency in English and their lack of skills (Birrell and Healy 2005). There is little prospect that this group will be capable of self funding their retirement needs (Birrell and Healy 2005).

The Second part of Section C relates to Housing. Broadly, the questions elicit information regarding:

- Length of residence at the present housing type/tenure;
- Whether the interviewee stayed or moved following the loss of the partner;
- If they stayed, reasons for staying, ability to cope with maintaining the property, what support formal and informal social support networks they had, affordability, and the impact of these factors on housing decisions;

 If they moved, reasons for moving, ability to cope with maintaining the property, what support formal and informal social support networks they had, affordability and the impact of these factors on housing decisions;

4.2 Future housing aspirations

This section of the research will not only look at housing costs and its impact on housing decisions, but is also based on the premise that an understanding and appreciation of non economic factors (such as preferences and aspirations) must also be considered. Financial, social, personal and sentimental values will be probed during the interviews.

We anticipate a variety of responses, depending on whether the loss of the partner occurred through divorce/separation or by death. Further, the responses will also be influenced by whether the interviewee owns their own home or is living in a rental property. The literature identifies various findings regarding these matters. For example:

- Separated/divorced or widowed people may have very different housing pathways. In the case of divorce or separation, the marital home will often be sold and the proceeds divided between the parties. In this case one or both parties may end in the rental market. In the alternative, one of the parties may remain in the marital home whilst the other moves to a rental property. In a situation where the parties were renting, then both parties are likely to remain in the rental market.
- In the case of home owners who have lost a partner, most, are unlikely to move unless there are compelling reasons to do so (Beer et al. 2006).
- Older people who live in private rental properties have been identified as having
 the greatest financial and housing needs. They are confronted with an array of
 problems ranging from financial hardship to the suitability, security and
 affordability of this tenure. (Beer et al. 2006) highlight that older people living in
 privately rented properties are far more likely to move when compared with
 home owners, and that such move does not necessarily occur by their own
 choice.
- (Jones et al. 2004) highlight that obtaining information on housing aspirations of older people is difficult and there are gaps in our understanding in terms of choices, needs and expectations. They point out that housing choices are affected by a range of social and economic factors which include: "patterns of family formation and dissolution, living arrangements, economic resources and personal characteristics" (Jones et al. 2004). They highlight key attributes of housing which are valued by older people and include in their list factors and values such as independence, affordability, security of tenure, safety, adaptability to future care, location, suitability, companionship and avoiding isolation, size, amenity and space (Jones et al. 2004).

There has been considerable concern as to the ability of elderly home owners to maintain their homes and gardens:

- Older home owners have been increasingly residing in post-war suburbs which have larger gardens. As older people become more frail, these homes can be increasingly difficult to maintain. This can be further aggravated by the fact that most older people are on lower incomes.
- And the family networks may also assist in the form of additional help, as time passes, the availability of family networks diminishes. This is because siblings are either in the workforce or may themselves require care (Beer et al. 2006).

The likelihood of increased care being required with age may influence older people's ability to reside in their own homes. (Beer et al. 2006).

The housing careers and needs of immigrants are likely to differ from the Australian population of Anglo Celtic origin:

- Home ownership amongst the older/more established immigrant communities is high whilst recently arrived immigrants are more likely to have less assets/resources and are likely to be living in rental properties (Beer et al. 2006).
- Older Australians from culturally and linguistically diverse backgrounds may deal with the experience of the loss of a partner through death or divorce in a different manner than those from Anglo Celtic backgrounds. This may be further aggravated by lack of proficiency in the English language and the absence of family networks.
- (Beer et al. 2006) highlights that "Cultural and attitudinal differences are transferred across generations and these effect how housing is consumed across the life course. In addition, the resources available to immigrants are likely to be less than those available non - immigrants, and this reflects the lower incomes of some groups of immigrants, variation in the levels of educational attainment, differences in family and household size, the presence or absence of specialised/bilingual community support"

The in-depth interviews will allow us to investigate all of the above areas. The interviews between the different cohorts (Divorcees / separated / widowed) will focus on three broad issues: (i) How aspirations/preferences differ between the different cohorts, (ii) how the aspirations /choices are changing between the cohorts and (iii) how the needs and issues affecting people from culturally and linguistically diverse backgrounds differ from the wider Anglo-Celtic community.

Section D: concerns networks, health and well being. In particular, the questions relate to:

- Recreational, social and friendships networks before the loss of their partner
- Whether these networks changed since loss of the partner
- Whether respondent currently belonged to any social groups and if so how often they would go to thee or see friends or family
- Overall health and whether deteriorated since loss of partner (including probing for sense of loneliness and isolation)

The overall aim of this section is to ascertain the impact of the loss of partner on social and friendship networks and the impact of these factors on housing and choices and well – being.

5 THE POLICY SIGNIFICANCE OF THIS RESEARCH

The number of older single persons is growing rapidly and this is expected to accelerate as the baby boomer generation ages. The increase is largely due to a higher divorce rate, a large increase in the number of older divorcees and declining remarriage rates.. Older single person households are more reliant on age related welfare services, including housing assistance and thus increasing numbers in this group are of particular significance from a policy perspective.

A finding of importance is the gender dimension to these issues. Over 70% of older singles that have lost a partner are women. This is because women live longer than men, and older men are more likely to repartner than older women. The disadvantaged housing market circumstances of older singles predominantly affects women. There are potential links here with other areas of policy concern, and in particular the access that older women have to their ex-partner's occupational pension and the importance of housing wealth to the well being of divorced women who do not remarry.

The median age, at which marriages fail has increased to 43 years for males and 40 for females, but it remains below 50. Thus the initial housing stress and need for housing assistance experienced during early middle age. The importance of this observation is that loss of a partner is not restricted to older persons (+50 years). For those experiencing household fracture due to divorce and separation, the initial consequences in terms of need for housing assistance will often occur during a person's middle years (30s and 40s). By restricting attentions to the 50+ age group we may be ignoring an important group whose demand for housing assistance is increasing. We intend to follow this up in more depth in the remainder of the project.

Loss of a partner has adverse consequences on a household's prospects of attaining or retaining homeownership status. This is particularly the case for those who are divorced and separated, but less so for widows. Widows typically inherit the estate of their partners at bereavement; this will often include the family home. Divorce and separation usually occurs earlier in the life course, and before the mortgage has been repaid. The loss of economies of scale in consumption, and benefits of specialisation can leave divorced and separated persons in precarious housing circumstances that make it difficult to retain homeownership. Our preliminary findings show this clearly. Our future panel data and qualitative research analysis will track housing pathways following divorce and separation. We expect our findings to shed insight into the types of housing assistance that might best assist housing wellbeing following divorce and separation.

Though there is a declining rate of remarriage <u>among divorces</u>, the majority nonetheless remarry. However it would seem that repartnered married couples are less likely to attain homeownership as compared to their continuously married counterparts, and have higher levels of mortgage debt as retirement age approaches. The policy focus is typically framed around older single persons that have lost a partner. Until recently, there has been an assumption is that repartnering reverses the adverse housing and financial consequences that follow household dissolution. The preliminary findings reported in this positioning paper question that assumption and warrant a wider policy perspective. Our panel data analysis will explore this further.

Our analysis of Housing Affordability Stress shows that older singles that have lost a partner pay gross housing costs that are typically a relatively higher proportion of their income than married couples have to pay. As a result older singles do express a higher demand for housing assistance. Among private renters the divorced and separated have a relatively high eligibility rate for CRA. However, even after taking CRA entitlements into account a little under 1 in 4 older singles that are divorced or

separated are in HAS. A relatively high proportion of the divorced and separated are also resident in public housing. The demographically driven projections for demand for public housing that will be conducted in the remainder of the project, will investigate whether this group's demand for housing assistance will increase or decrease in the future. Early findings (see section 3.3.3) indicate a projected increase in the numbers of female sole parents entering public housing waiting lists.

APPENDIX

The Implications of Loss of a Partner on the Housing Pathways of Older People

Screening Form

The xxx is carrying out a project on the housing and financial consequences for people aged 50 and older who have lost their partner. We want to interview a cross section of people, so could you give us a few details about yourself. **Please circle the correct answer to the following questions:**

Gender:	Male / Female
Age:	50-59 / 60-74 / 75 or older
Reason for loss of partner:	Death / Divorce / Separation
Have you lost your partner in the last three years:	Yes / No
Your occupation:	Retired / Employed Full Time/ Employed Part time
Income source:	Pension / Employment / Other:
Current housing type:	House/ flat/ unit/ caravan / other:
Tenure type:	Home owner/ home purchaser/ private renter/ public renter/ other tenure
Country of birth:	type:
Language Spoken at Home:	
Aboriginal/ATSI	
I am willing to take part in the xx research on peophusband/wife/partner in the last three years. I und hour and that I will be paid \$30 for my time. I undearrange an interview at a convenient time.	erstand that the interview will take about one
Signature:	
Name:	
Address:	
Telephone:	
Date:	

Appendix

The Implications of Loss of a Partner for Older People

Introduce the project: My name is I am from the xx at xx University. We are carrying out a project on the housing and financial consequences for people aged 50 and older who have lost their partner. If there are any questions that you would prefer not to answer, just let me know. I am going to tape record the interview so I have an accurate record of what you say, but there will not be any personal details about you in the report.

SECTION A: NATURE OF THE RELATIONSH	
1. Were you married or in a de-facto relationship?	Nature of relationship
	i. married ii. defacto
2. How long was the marriage / relationship?	<u>Length of relationship</u>
	i. 0-10 years
	ii. 11-20 years
	iii. 21 -30 years
	iv. 31-40 years
	v. 41 or more
3. Do you have any children?	Children of the relationship
	i. Yes
	ii. No
(If so, how many and how old are they?)	No of children: 0 1 2 3 4 5 6
Do any of them live with you?	i. No Children at home
	ii. YesChildren at home

SECTION B: CIRCUMSTANCES SURROUNDING LOSS OF PARTNER The next group of questions relate to the circumstances surrounding the loss of your partner 4. What were the circumstances of the loss of your partner? Reason for loss Death i. ii. Divorce Separation 5. When did it occur? When loss occurred 1 yr ii. 2 yrs iii. 3yrs Other..... 6. How did you cope emotionally at the time of the loss? Level of trauma (time of death) (Probe: form of support provided and by whom) Very high ii. High iii. Medium Low iv. Nil v. Other: vi. 7. How are you coping with the loss now? Level of trauma now Very high i. ii. High Medium iii. iv. Low Nil vi. Other: vii.

SECTION C: CHANGES TO LIFE AFTER THE LOSS OF PARTNER

I'm now going to ask a series of questions on the main the main changes to your life after the loss of your partner.

the loss of your partner.		
FINANCIAL:		
8. Had you and your former finances planned your finances	Retirer	ment plans before loss
for retirement?	of part	<u>ner</u>
Probe as to: Former partner's superannuation	i.	Former partner's super
Respondent's superannuation	ii.	Respondent's super
Self funded retiree	iii.	Savings
Savings	iv.	Income from
		assets/property
Income from assets/property	V.	Govt pension
Government pension	vi.	Other:
How they were going to financially manage their retirement		
If already retired, probe as to how they were managing to cope.		
9. What is your current source of income?	Curren	t income
	i.	Super (both parties)
	ii.	Super (former partner)
	iii.	Super (respondent's)
	iv.	Income from
		assets/property
	V.	Savings
	vi.	Government pension
	vii. 	Employment (FT / PT)
	viii.	Other:
10. How much has your income changed since the loss of	<u>Fina</u>	ncial consequence of
your partner?		<u>los</u> s
(Probe: as to whether respondent is financially better/worse off or	i.	Worse off
about the same)		
	ii.	Better
If Possible - Quantify	iii.	About same
	iv.	D/K
	V.	Other
11. How much have your assets changed since the loss of	Chanc	ge in assets
your partner? Probe as to whether changes in household assets have	j,	Assets have
influenced housing decisions		increased
<u> </u>	ii.	Assets have
		decreased
	1	300.0000

	- .	
	iii	ii. About the same
	iv	v. D/K
	\rightarrow \nu	v. Other
12. How do you expect to pay for your future needs?	<u>P</u>	Payment for future needs
(Probe: eg for retirement village, holidays etc.)	i.	Through current
		income source
(Probe: do they expect to use up all their assets while they are	ii.	Sale of
alive)		assets/property
	iii.	Borrowing
	iv.	D/K
	V.	Other:
HOUSING:		
13. How long have you lived in your present house/flat?		Length at current
		<u>residence</u>
		iyears
14. Is this where you lived when you lost your partner?	\longrightarrow	Whether respondent
14. 15 tillo miloto jou intou milot jou look jour parane.		moved since they lost
		their partner
		i. Moved
		ii. Did not move
		···
Instruction:		
If the person stayed, go to Question 15		
If the person moved, go to Question 22		
If they are in the same house/flat		
15. What sort of housing do you live in (Tenure Type)?	Tenu	ure type
	Hou	se, flat, unit, caravan,
	bunç	galow,
	Othe	er:
16 Do you own it or are you renting?	<u> </u>	wner: outright

If the person stayed, go to Question 15		
If the person moved, go to Question 22		
If they are in the same house/flat		
15. What sort of housing do you live in (Tenure Type)?		Tenure type
		House, flat, unit, caravan,
		bungalow,
		Other:
16. Do you own it or are you renting?	i.	Owner: outright
If renting: from whom are you renting from?	ii.	Owner: mortgage
If home owner: do you have a mortgage?	iii.	Rent : private
	iv.	Rent: public
	V.	Other
17. Why did you decide to stay?		Why stoyed
		Why stayed
(Probe also as to person's attachment to the location as	i.	Happy memories/sentimental
	_	

opposed to the property)		
	ii.	Location
	iii.	Friends/networks
	iv.	Could not afford to move
	V.	Convenient
	vi.	Thought of moving& did not
		as
	vii.	D/K
	viii.	Other
18. Is this house/flat appropriate for you in terms of size/location/state of repair?		Level of satisfaction
i. (Probe: whether they would consider making	ii.	
changes and why)		
iii.	iv.	
V.		
	vi.	Approp. In terms of:
	vii.	Not approp. In terms of:
	viii.	Satisfaction: VH, H, M, L
	ix.	Other
19. Can you afford the property?		<u>Affordability</u>
		Affordable
(Probe: If not affordable, probe at so why not and what	i.	Allordable
(Probe: If not affordable, probe at so why not and what experiences they are having with payments)	-	
· ·	ii.	Not affordable
· ·	ii. iii.	Not affordable Get by
experiences they are having with payments)	ii.	Not affordable
· ·	ii. iii. iv.	Not affordable Get by
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends,	ii. iii. iv.	Not affordable Get by Other:
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC)	ii. iii. iv. <u>At</u>	Not affordable Get by Other: bility to cope with property Able to cope on own
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small	ii. iii. iv.	Not affordable Get by Other: billity to cope with property Able to cope on own Not able to cope on own.
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC)	ii. iii. iv. <u>At</u> i.	Not affordable Get by Other: billity to cope with property Able to cope on own Not able to cope on own. Assistance from:
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small	ii. iii. iv. AL ii.	Not affordable Get by Other: Dility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small	ii. iii. iv. <u>At</u> i. ii. iiv.	Not affordable Get by Other: bility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small	ii. iii. iv. At. ii. iii. iv. v.	Not affordable Get by Other: Dility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small	ii. iii. iv. Al. ii. iv. v. vi.	Not affordable Get by Other: bility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children Friends
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small repairs/maintenance/cleaning)	ii. iii. iv. Al. ii. iv. v. vi. vii.	Not affordable Get by Other: bility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children Friends Other
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small repairs/maintenance/cleaning)	ii. iii. iv. Al. ii. iv. v. vi. vii.	Not affordable Get by Other: Dility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children Friends Other
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small repairs/maintenance/cleaning) 21. Do you want to move in the future? (Probe retirement village, living with children, downsize,	ii. iii. iv. Al. ii. iv. v. vi. vii.	Not affordable Get by Other: bility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children Friends Other
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small	ii. iii. iv. Al. ii. iv. v. vi. vii.	Not affordable Get by Other: bility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children Friends Other onsideration of future move Will not consider:
20. Are you able to cope with looking after the property? (Probe: whether they get help from children, friends, HACC) (Probe as to ability to maintain garden, carry out small repairs/maintenance/cleaning) 21. Do you want to move in the future? (Probe retirement village, living with children, downsize,	ii. iii. iv. Al. ii. iv. v. vi. vii.	Not affordable Get by Other: Dility to cope with property Able to cope on own Not able to cope on own. Assistance from: HACC Family Children Friends Other

iii.	Friends/networks
iv.	Could not afford to move
v.	Convenient
vi.	D/K
vii.	Other
 	Will consider:
viii.	Retirement village
ix.	With children
x.	Downsize
хi.	Closer to children/friends
xii.	Closer to transport/shops

Instruction: Go to Question 24		
Instruction: Go to Question 31 If they moved since loss of partner		
22. Where did you live before the loss of your partner?	Teni	ure type before loss of
	1011	partner
	H	ouse, flat, unit, caravan,
		ingalow, other:
23. Did you own it or were you renting?	i.	Owner: outright
		Ŭ
If renting: from whom were you renting from?	ii.	Owner: mortgage
If home owner: did you have a mortgage?	iii.	Rent : private
, 5 5		·
	iv.	Rent: public
	v.	Other
24. Why did you move?		
	Reaso	ns for move after loss
(Probe: financial/emotional factors:- eg house too big/too small or downsize to release money, move to a better location.	i.	Downsize
Live with or closer to family, emotional reasons, health factors	ii.	Health reasons
etc).		
(Probe also as to person's attachment to the location as	iii.	Closer to
opposed to the property)		friends/children
	iv.	Emotional factors
	v.	Closer to amenities
	vi.	Retirement village
	vii.	Financial
	viii.	Other:
25. Was the move a success?	Wheth succe	er move was ssful
	- i.	Yes
	1	

	iii.	Mixed
	iv.	D/K
	V.	Other
26. Now some questions about you current property	Tenur partne	e type after loss of er
What type of housing do you currently live in?	House	, flat, unit, caravan,
	bunga	low, other:
27. Do you own it or are you renting?	i.	Owner: outright
If renting: from whom are you renting from?	ii.	Owner: mortgage
If home owner: do you have a mortgage?	iii.	Rent: private
	iv.	Rent: public
	V.	Other:
28. Is this housing appropriate for you in terms of size/location/state of repair?		of satisfaction with
(Probe: whether they would consider making changes and why)	housii i.	Approp. In terms of:
	ii.	Not approp. In terms
		of:
	iii.	Satisfaction: VH, H, M,
		L
	iv.	Other
29. Can you afford the property?		
	Afford	<u>lability</u>
(Probe: If not affordable, probe at so why not and what experiences they are having with payments)	i.	Affordable
	ii.	Not affordable
	iii.	Get by
	iv.	Other:
30. Are you able to cope with looking after the property?	<u>Ability</u>	to cope with property
(Probe: whether they get help from children , friends, HA	i.	Able to cope on own
(Probe as to ability to maintain garden, carry out small	ii.	Not able to cope on
repairs/maintenance/cleaning)		own. Assistance from:
	iii.	HACC
	iv.	Family
	V.	Children
	vi.	Friends
	vii.	Other
31. Do you want to move in the future?	Cons	ideration of future move
(Probe retirement village, living with children, downsize, closer	1	Will not consider:
to shops/transport)		

memories/sentimental ii. Location iii. Friends/networks iv. Could not afford to move v. Convenient vi. D/K vii. Other Will consider: viii. Retirement village ix. With children x. Downsize
iii. Friends/networks iv. Could not afford to move v. Convenient vi. D/K vii. Other Will consider: viii. Retirement village ix. With children
iv. Could not afford to move v. Convenient vi. D/K vii. Other Will consider: viii. Retirement village ix. With children
move
v. Convenient vi. D/K vii. Other Will consider: viii. Retirement village ix. With children
vi. D/K vii. Other Will consider: viii. Retirement village ix. With children
vii. Other Will consider: viii. Retirement village ix. With children
Will consider: viii. Retirement village ix. With children
viii. Retirement village ix. With children
ix. With children
x. Downsize
xi. Closer to
children/friends
xii. Closer to
transport/shops
SECTION D: NETWORKS, HEALTH & WELL BEING
<u>Networks</u>
32. Have you re- partnered? Whether re-partnered
since the loss
i. Yes
ii. No
33. What sort of recreational, social and friendship networks <u>Networks during</u>
did you have before you lost you partner? <u>relationship</u>
i. Extensive social
networks
ii. Average social
networks
iii. Minimal social
networks
iv. No social networks
v. Did/did not
participate in social
and recreational
activities
vi. Other:
34. Have these changed since you lost your partner? Change in networks after loss of partner
i. No change
ii. Minimal change
iii. Substantial change
iv. Other

35. Do you belong to any social groups?	Membership to social group(s)
cor be year never great and great gr	member of the obtaining output

If so, how many????	i.	Yes
	ii.	No
	iii.	Other
36. How often do you go to them?	<u>Fr</u>	requency in visiting social
	gı	oups
	i.	Daily
	ii.	Weekly
	iii.	Monthly
	iv.	Other
37. How often do you see friends or do activities	Freque	ency of visiting/meeting
together?	with fr	<u>riends</u>
	i.	Daily
	ii.	Weekly
	iii.	Monthly
	iv.	Other
38. How often do you see family or do activities	Fred	quency of visiting/meeting
together?		with family
	i.	Daily
	ii.	Weekly
	iii.	Monthy
	iv.	Other
Health		
Health		
	Curi	rent health and well being
39. Can you describe your overall health and well being.	Curi	rent health and well being
39. Can you describe your overall health and well being.	<u>Curi</u>	rent health and well being
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	<u>Curi</u>	rent health and well being
39. Can you describe your overall health and well being.		rent health and well being Good
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	Curi i. ii.	Good
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	i.	
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	i. ii.	Good Average
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	i. ii. iii.	Good Average Bad
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	i. ii. iii. iv.	Good Average Bad Deteriorating
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness	i. ii. iii. iv. v. vi.	Good Average Bad Deteriorating Very bad
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation)	i. ii. iii. iv. v. vi.	Good Average Bad Deteriorating Very bad Other:
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner?	i. ii. iii. iv. v. vi.	Good Average Bad Deteriorating Very bad Other:
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness	i. ii. iii. iv. v. vi.	Good Average Bad Deteriorating Very bad Other:
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation)	i. ii. iii. iv. v. vi.	Good Average Bad Deteriorating Very bad Other:
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation) Probe as to impact of housing and housing wealth on well	i. ii. iv. v. vi. Deter	Good Average Bad Deteriorating Very bad Other: ioration in health after loss of partner
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation) Probe as to impact of housing and housing wealth on well being Probe as to impact of health and on well being on	i. ii. iv. v. vi. Deter	Good Average Bad Deteriorating Very bad Other: ioration in health after loss of partner
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation) Probe as to impact of housing and housing wealth on well	i. ii. iv. v. vi. Deter i.	Good Average Bad Deteriorating Very bad Other: ioration in health after loss of partner
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation) Probe as to impact of housing and housing wealth on well being Probe as to impact of health and on well being on	i. ii. iv. v. vi. Deter i.	Good Average Bad Deteriorating Very bad Other: ioration in health after loss of partner Yes
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation) Probe as to impact of housing and housing wealth on well being Probe as to impact of health and on well being on	i. iii. iv. vi. Deter i.	Good Average Bad Deteriorating Very bad Other: ioration in health after loss of partner Yes No Deteriorating
39. Can you describe your overall health and well being. (Probe: physical and emotional including sense of loneliness and isolation) 40. Has your overall health deteriorated since the loss of your partner? (Probe: physical and emotional including sense of loneliness and isolation) Probe as to impact of housing and housing wealth on well being Probe as to impact of health and on well being on	i. ii. iv. v. vi. Deter i.	Good Average Bad Deteriorating Very bad Other: ioration in health after loss of partner Yes

Thank you for your help.

SECTION E: BACKGROUND INFORMATION

This section is to be completed by the Interviewer immediately at the end of the Interview

Gender:	Male / Female
Age:	50-59 / 60-74 / 75 or older
Reason for loss of partner:	Death / Divorce / Separation
Have you lost your partner in the last three years:	Yes / No
When loss of partner occurred: (other)	1 year ago / 2 years ago / 3 years ago /
Employment Status:	Retired / Employed Full Time/ Employed Part time
Income source:	Pension / Employment / Other:
Current housing type: Other:	House/ flat/ unit/ caravan /
Tenure type at loss of partner: renter/ public renter/	Home owner/ home purchaser/ private
	other tenure
Country of birth:	type:
Language Spoken at Home	
Interviewer's Comments:	

REFERENCES

- ABS (2006). Australian Historical Population Statistics. Cat No. 3105.0.65001, Australian Bureau of Statistics.
- AHURI (1996). Statistical Analysis of Older People and Their Housing Circumstances. New Homes for Old Program. Canberra, Commonwealth Department of Transport and Regional Development, AGPS.
- AHURI (2004). "Housing Futures in an Ageing Australia." *AHURI Research and Policy Bulletin*(43): 1-8.
- AMP NATSEM (2002). Live Long and Prosper?: The Income and Wealth of Those About to Retire. *Income and Wealth Report*, AMP NATSEM.
- AMP NATSEM, Kelly, S. and Harding, A. (2005). Love Can Hurt, Divorce Will Cost. *Income and Wealth Report Issue 10*, AMP-NATSEM.
- Beer, A., Fauldner, D. and Gabriel, M. (2006). 21st Century Housing Careers and Australia's Housing Future: Literature Review. *AHURI Positioning Paper*, AHURI.
- Birrell, B. and Healy, E. (2005). "Baby Boomer Affluence Myth or Reality?" *Just Policy* 37: 33 40.
- Boheim, R. and Taylor, M. P. (2000). "My Home Was My Castle: Evictions and Repossessions in Britain." *Journal of Housing economics* 9: 287-319.
- Brooke, L., Davidson, S. and Kendig, H. (1998). Support Needs of Older People in High Rise Public Housing. Melbourne, Victorian Department of Human Services.
- Clarke, W. A. V., Deurloo, M. C. and Dieleman, F. M. (2003). "Housing Careers in the United States, 1968-'93: Modelling the Sequencing of Housing States." *Urban Studies* 40(1): 143-160.
- Costello, P. (2002). Intergenerational Report 2002-03. 2002-03 Budget Paper No. 5. Canberra, Commonwealth of Australia.
- Dieleman, F. M., Clark, W. A. V. and Duerloo, M. C. (1995). "Falling out of the Home Owner Market." *Housing Studies* 10(1): 3-15.
- Evandrou, M., Falkingham, J. and Scott, A. (2001). The Dynamics of Living Arrangements in Later Life: Preliminary Findings. *Sage Discussion Paper No. 4*. London, ESRC Sage Research Group, London School of Economics.
- Faulkner, D. (2001). Linkages among Housing Assistance, Residential (Re)Location and Use of Community Health and Social Care among Old-Old Adults: Shelter and Non-Shelter Implications for Housing Policy Development. Melbourne, Australian Housing and Urban Research Institute.
- Feijten, P. and Mulder, C. H. (2005). "Life Course Experience and Housing Quality." Housing Studies 20(4): 571 - 87.
- Fiedler, J. (1999). "An Age Old Demand for Dignity." Parity 12(9): 14.
- Fischer, P. A. and Malmberg, G. (2001). "Settled People Don't Move: On Life Course and (Im-) Mobility in Sweden,." *International Journal of Population Geography* 7: 357 371.
- Flatau, P., Hendershott, P., Watson, R. and Wood, G. (2004). An Examination of the Role of Labour Market, Social and Economic Determinants. *Final Report*, AHURI.

- Grossbard Shechtman, S. (1993). On the Economics of Marriage: A Theory of Marriage, Labor and Divorce, Westview Press, USA.
- Grossbard Shechtman, S. (2003). *Marriage and the Economy: Theory and Evidence from Advanced Industrial Societies*. Cambridge, Cambridge University Press.
- House of Representatives Standing Committee on Health and Ageing (2005). Future Ageing: Report on a Draft Report of the 40th Parliament: Inquiry into Long-Term Strategies to Address the Ageing of the Australian Population over the Next 40 Years. Canberra, Parliament of the Commonwealth of Australia.
- Howe, A. (2003). Housing an Older Australia: More of the Same or Something Different? Keynote Address to the Housing Futures in an Ageing Australia. AHURI and the Myer Foundation Conference, Melbourne.
- Howe, A. L. (1992). Housing for Older Australians: Affordability, Adjustments and Care,. *National Housing Strategy, Background paper*. Canberra. 8.
- Jones, A., Bell, M., Tilse, C. and Earl, G. (2004). Rental Housing Provision for Lower Income Older Australians, AHURI, Queensland Research Centre.
- Kendig, H. (1984). "Housing Careers, Life Cycle and Residential Mobility: Implications for the Housing Market." *Urban Studies* 21: 271 283.
- Kendig, H. (1990). A Life Course Perspective on Housing Attainment. *Housing Demography: Linking Demographic Structure and Housing Markets*. D. Myers. Madison Wisconsin, University of Wisconsin Press: 133-156.
- Kendig, H. and Neutze, M. (1999). *Housing Implications of Population Ageing in Australia*. Policy implications of the ageing of Australia's population: conference proceedings,, Productivity Commission and Melbourne Institute of Economic and Social Research, Commonwealth of Australia.
- Lehrer, E. (2003). The Economics of Divorce. *Marriage and the Economy: Theory and Evidence from Advanced Industrial Societies*. S. Grossbard Shechtman. Cambridge, Cambridge University Press.
- McCarthy, P. and Simpson, B. (1991). *Issues in Post Divorce Housing*, Avebury, England.
- McDonald, P. (2003). Transformations in the Australian Family'. *The Transformation of the Australian Population: 1970-2030.* S. Khoo and P. McDonald. Sydney, University of New South Wales Press: 77-103.
- McDonald, P. and Kippen, R. (1999). *Ageing: The Social and Demographic Dimensions*. Policy implications of the ageing of Australia's population, Productivity Commission and Melbourne Institute of Economic and Social Research, Commonwealth of Australia.
- Morris, A., Judd, B. and Kavanagh, K. (2005). "Marginality Amidst Plenty: Pathways into Homelessness for Older Australians." *Australian Journal of Social Issues* 40(2): 241-251.
- Olsberg, D. (2005). "Women, Superannuation and Retirement: Grim Prospects Despite Policy Changes." *Just Policy* 35: 31-38.
- Olsberg, D., Perry, J., Encel, S. and Adorjánÿ, L. (2004). Ageing-in-Place? Intergenerational and Intra-Familial Housing Transfers and Shifts in Later Life,. Sydney, AHURI, UNSW-UWS Research Centre.
- Olsberg, D. and Winters, M. (2005). Ageing in Place: Intergenerational and Interfamilial Housing Transfers and Shifts in Later Life, AHURI.

- Sevak, P., Weir, D. and Willis, R. J. (2003/4). "The Economic Consequences of a Husband's Death: Evidence from the Hrs and Ahead." *Social Security Bulletin* 65(3): 31-44.
- Smyth, B. and Weston, R. (2000). Financial Living Standards after Divorce: A Recent Snapshot, Australian Institute of Family Studies.
- Winter, I. and Stone, W. (1999). Home Ownership: Off Course? *Australia's Housing Choices*. Y. J and M. Wulff. Brisbane, University of Queensland Press.: 43-52.
- Wood, G. A., Forbes, M. and Gibb, K. D. (2005). "Direct Subsidies and Housing Affordability in Australian Private Rental Markets." *Environment and Planning C: Government and Policy* 23: 759-783.
- Zagorsky, J. (2005). "Marriage and Divorce's Impact on Wealth." *Journal of Sociology* 41(4): 406-424.

AHURI Research Centres

Queensland Research Centre
RMIT-NATSEM Research Centre
Southern Research Centre
Swinburne-Monash Research Centre
Sydney Research Centre
UNSW-UWS Research Centre
Western Australia Research Centre



Australian Housing and Urban Research Institute Australian Housing and Urban Research Institute

Level 1 114 Flinders Street, Melbourne Victoria 3000

Phone +61 3 9660 2300 Fax +61 3 9663 5488

Email information@ahuri.edu.au Web www.ahuri.edu.au